

Oldenburg Discussion Papers in Economics

More Than Money: The Critical Role of Management in Educational Aid Effectiveness in Africa

Abigail Opokua Asare

V – 449-25

March 2025

Department of Economics University of Oldenburg, D-26111 Oldenburg

More Than Money: The Critical Role of Management in Educational Aid Effectiveness in Africa *

Abigail Opokua Asare[†]

Carl von Ossietzky University Oldenburg

March 19, 2025

Abstract

The effectiveness of educational aid in Africa is a pressing issue, with little consensus on whether the management quality of aid has significant contribution toward achieving inclusive, equitable, and quality education on the continent by 2030. Despite substantial inflows of educational aid from both bilateral and multilateral sources, Africa continues to report the world's highest illiteracy rates, indicating potential inefficiencies in educational aid management. This paper investigates whether the impact of World Bank–funded educational projects across Africa on literacy rates depends on the quality of project management. The findings reveal that educational projects managed in a highly satisfactory manner significantly reduce illiteracy, regardless of the quantity of aid or volume of aid disbursement. Meanwhile, projects managed in a sub-satisfactory manner show no progress at all. These findings highlight that effective management is far more critical to success than the amount of aid provided and suggest that reforming management practices, policymakers and international organizations could improve the effects of educational aid, offering a targeted strategy to drive Africa's educational progress.

Keywords: aid effectiveness, management quality, illiteracy, DHS, World Bank, Africa

JEL Classification Numbers: F35, O12, O55, I2, I21

^{*}I would like to sincerely thank Jürgen Bitzer, Erkan Gören, Bernhard C. Dannemann, and seminar participants at the Carl von Ossietzky Universität Oldenburg, 2024, for helpful comments and suggestions. All remaining errors are my own. I have no known conflict of interest to disclose.

[†]Carl von Ossietzky Universität Oldenburg, School of Computing Science, Business Administration, Economics, and Law (Faculty II), Institute of Economics, Building A5, 26129 Oldenburg, Germany, Tel.: +49-441-798-2625, e-mail: abigail.asare@uol.de.

1 Introduction

Worldwide, there is a "learning crisis" (Stacy et al., 2023; UNICEF, 2024) driven by factors such as low investments in education, economic hardship, conflicts, and climatic conditions (e.g., flooding and drought) that hinder educational development. The crisis is particularly acute in Africa. Despite efforts to achieve universal primary education and quality education for all under United Nations Millennium Development Goal 2 (MDG 2) and Sustainable Development Goal 4 (SDG 4)¹, high levels of educational exclusion persist in Africa (UNESCO, 2024a). In Sub-Saharan Africa, for example, around 60% of youths aged 15 to 17 are not in school, and roughly 48 million young people in this age group lack basic reading and writing skills (UNESCO, 2024a). Moreover, about 89% of 10-year-old children in Eastern and Southern Africa cannot read a simple text (World Bank, 2024). Orphans, children from rural areas, and people with disabilities are among the disadvantaged groups that experience educational exclusion (Lewin, 2020; Riddell & Niño-Zarazúa, 2016).

The World Bank has sought to address these challenges and accelerate the realization of SDG 4 by providing substantial financial aid for education in Africa. As of 2023, the World Bank's educational aid in Eastern and Southern Africa alone amounted to about \$6.33 billion across 35 projects (World Bank, 2024)². Although we expect these financial commitments to improve educational outcomes significantly, evidence suggests that project performance is often impeded by management challenges. This shows the importance of understanding how the quality of project management affects the success of educational aid and, by extension, the effectiveness of these projects in reducing illiteracy.

One strand of the literature on aid effectiveness and educational outcomes has focused on the relationship between educational aid and improvements in achievement metrics such as completion rates, revealing significant but varying impacts (Birchler & Michaelowa, 2016; d'Aiglepierre & Wagner, 2013; Dreher et al., 2008; Michaelowa & Weber, 2008; Michaelowa, 2004; Miningou, 2019). In a paper closely related to the present one, Yogo (2017) found that foreign aid positively influenced primary school completion rates in 35 Sub-Saharan African countries, but that the effect was relatively small. Their conclusion was that aid levels should be doubled to achieve universal primary education. Thierry and Emmanuel (2022) extended this analysis by assessing the role of financial development in increasing enrollment at all educational levels through structural economic factors (a country's financial system ³) such as accessibility and efficiency,

¹MDG 2 is aimed to achieve universal primary education; SDG 4 is dedicated to ensuring inclusive and equitable education and promoting lifelong learning opportunities for all.

²From 1995 to 2014, the number of aid projects in Africa was about 1,970 out of 5,684 geocoded projects (5,881 total projects) in 6,591 out of 61,243 project locations worldwide (AidData, 2017).

³Thierry and Emmanuel (2022) use nine indices: overall financial development, financial institutions, financial institution depth, financial institution access, financial institution efficiency, financial markets, financial market depth, financial market

and argued for strengthened domestic financial systems to ensure inclusive educational progress. Musah et al. (2024), in contrast, found that although investments in primary and tertiary education do improve some aspects of educational quality, secondary education continues to face challenges despite increased funding.

The aforementioned studies rely on country-level data, which obscures critical within-country variation by failing to control for unobserved heterogeneity (factors such as historical relations that might affect foreign aid allocation in a given geographical area) across countries. The present paper addresses this problem by employing a sub-national focus, analyzing geocoded data across multiple recipient countries to generalize findings more broadly. The empirical design enables identification of local variation, which is particularly relevant in Africa, where diverse communities face distinct educational challenges that affect long-run educational development. The focus in this paper—in contrast to Thierry and Emmanuel's emphasis on financial systems or Yogo's focus on aid allocation for primary education—is on the management quality of World Bank– funded projects as a determinant of illiteracy outcomes.

Another paper focusing on the sub-national level, but with a much narrower geographic scope than the present one, is that of Helskog (2024), who investigated the localized effects of development aid on primary school dropout rates in conflict-affected areas of Rwanda. Using a linear probability model to analyze over 63,000 individuals within 10 km of aid projects, the author found that development aid reduces the probability of dropout in genocide-affected areas by 20%, particularly in low-income households. While their study provides valuable insights, it also has limitations: It does not consider the quality of project management, which could significantly influence aid project effectiveness. It further raises questions about the generalization of findings to other contexts beyond Rwanda, given the country's unique post-genocide history. Moreover, it has difficulty isolating the specific impact of educational aid since only 12 of the 199 projects studied directly targeted education. To address this, Helskog takes a broad approach, showing how forms of non-educational aid, such as health and agriculture initiatives, indirectly benefit education by improving household incomes and helping families keep their children in school. While this sheds light on the far-reaching effects of development aid, it also shows the need for further research focusing on how targeted educational aid and effective management can make a difference. The present paper seeks to answer this question by not only examining conflict-affected areas but also looking closely at managerial aspects of aid allocation and the specific contributions of the targeted educational aid in reducing the illiteracy rate in a given spatial unit.

A second strand of research analyses the determinants of World Bank project management performance, focusing on country-level factors such as GDP growth, political stability, and institutional quality (Ashton

access, and financial market efficiency to proxy for financial development in a given country.

et al., 2022; Briggs, 2019; Chauvet et al., 2010; Denizer et al., 2013; Dreher et al., 2013; Guillaumont & Chauvet, 2001; Guillaumont & Laajaj, 2006), as well as project-specific characteristics such as the assignment and supervision of task team leaders (TTL) (Heinzel & Liese, 2021; Honig et al., 2022; Kilby, 2000; Limodio, 2021) as critical for overall aid success. However, this research largely overlooks how management performance directly influences specific intended outcomes such as reductions in illiteracy rates.

No study to date, to the author's knowledge, has combined these two strands of literature by investigating how the quality and efficiency of management in World Bank–funded educational projects translates into educational outcomes at the sub-national level. The present paper makes the unique contribution of analyzing the impact of project management quality—measured by World Bank project ratings—on illiteracy rates at the sub-national level in Africa. It examines distinctions between projects rated highly satisfactory (HS), moderately satisfactory (MS), and less satisfactory (LS) to provide nuanced insights into the differential effects of project management on educational development. Furthermore, it applies spatial buffers to match World Bank–funded educational projects with educational information. This offers a more precise understanding of how management quality influences educational outcomes in specific areas by uncovering granular patterns of aid effectiveness. Additionally, it investigates how the management quality of one project affects that of other projects in the same spatial unit, addressing the question of whether an HS project can improve the impact of an LS educational project in the same cluster. This approach fills critical gaps in the literature by linking the management quality of diverse project clusters to sub-national educational outcomes, offering a novel perspective on optimizing aid strategies for educational improvement in Africa.

In this paper, I match the sub-national geocoded data on World Bank project management quality (proxied by project ratings) produced by the Internal Evaluation Group (IEG) from AidData (AidData, 2017) with geocoded data on education from the Demographic and Health Surveys (DHS) Program (ICF, 2024a) in 33 African countries. Using data from 1992 to 2020 and a 50 km buffer as baseline results, my results show that educational projects with HS management significantly reduce illiteracy rates, while the opposite holds for MS and LS projects. Probing further reveals management quality dependencies. Areas with HS projects show consistently reduced illiteracy, whereas areas with MS and LS projects show increased illiteracy. This suggests that ineffectively managed educational aid can actually worsen educational outcomes in a given spatial unit. Results also show that in both urban and rural areas, HS projects consistently reduce illiteracy, with greater impacts in urban areas. In urban areas, MS projects worsen outcomes when implemented alone but improve outcomes when paired with HS projects is not observed. LS

projects, on the other hand, consistently exacerbate illiteracy, even when paired with MS or HS projects. Robustness checks based on different buffer sizes show that the choice of distance around the DHS program location does not drive the results. These results suggest that to maximize impact, policies must prioritize scaling HS projects and fostering effective project combinations. Rural areas need strategic investments in management capacity and tailored project designs to address persistent challenges.

The structure of the paper is as follows: In Section 2, I describe the data and key variables used for the analysis. Section 3 focuses on the empirical and econometric framework. Section 4 presents the empirical results, Section 5 discusses the robustness checks, Section 6 discusses the paper's findings, and Section 7 concludes.

2 Data

In this analysis, I combine georeferenced aid data from AidData (AidData, 2017), which reports on World Bank aid projects in each recipient country in a given year. It spans the period 1995 to 2014. The educational data are obtained from the DHS survey program (ICF, 2024b), which covers a large number of years, 1992 to 2020. The analyses presented here span that entire period, encompassing approximately 57,340 clusters with little or no completed educational aid projects prior to the survey date. Further details on the construction of dependent and explanatory variables are presented in the following subsections 2.1, 2.2, 2.3.

2.1 Dependent Variable

The "share of illiteracy" is the dependent variable based on geocoded data from DHS (ICF, 2024a) for 33 countries between 1992 and 2020 (see Appendix Tables A.8 for countries and surveys included). The DHS data are nationally representative and include demographic, educational, and geographic information for households and individuals. Educational information includes and is not limited to years and level of completed education and current schooling status. Combining this information, I calculate the proportion of individuals aged 6 to 24 years without formal or completed primary education. The illiteracy rate in a cluster is expressed mathematically as:

$$illiteracy_{dct} = \frac{\sum S_{idct}}{\sum N_{idct}} \tag{1}$$

where: $illiteracy_{dct}$ indicates the percentage share of people in the 6 to 24 age group within cluster d in country c at time t who have no formal or completed primary education. $\sum S_{idct}$ is the total number of individuals with no formal education or no completed primary schooling in that age group.

Statistics	Obs.	Mean	St. Dev.	Min	Max		
	Dependent va	ariable					
Share of illiteracy (6-24 years)	58,182	24.408	24.895	0	100		
	Completed ai	d projects – i	ndicator				
Dummy: highly satisfactory edu.	58,182	0.175	0.380	0	1		
Dummy: moderately satisfactory edu.	58,182	0.252	0.434	0	1		
Dummy: less satisfactory edu.	$58,\!182$	0.057	0.232	0	1		
	Completed ai	d projects – d	counts				
Total completed aid	58,182	9.175	15.604	0	123		
Total completed edu. aid	58,182	1.673	3.933	0	46		
Highly satisfactory, completed edu.	58,182	0.512	1.914	0	23		
Moderately satisfactory, completed edu.	58,182	0.559	1.379	0	18		
Less satisfactory, completed edu.	58,182	0.088	0.499	0	9		
	Completed ai	d projects – l	og of disburs	ement			
ln(Value edu.)	21,038	15.775	1.501	11.593	19.845		
ln(Highly satisfactory)	10,204	15.946	0.975	13.045	18.673		
ln(Moderately satisfactory)	$14,\!651$	15.355	1.500	11.593	19.841		
ln(Less satisfactory)	3,318	14.315	1.537	12.109	18.269		
	Geographic as	nd weather co	ontrols				
Num. conflicts 5 years prior	58,182	4.711	18.824	0	573		
Population per 100,000	58,182	21.293	32.300	0	244.025		
Mean nighttime light	58,182	7.713	8.674	2.259	59.537		
Mean precipitation	$58,\!182$	89.529	57.391	0.009	289.387		
Mean temperature	58,182	23.847	3.659	10.115	31.030		
Mean drought	58,182	-0.287	0.714	-2.456	2.743		
DHS Variables							
Average age	58,182	22.996	3.744	12.500	62.233		
Average household size	$58,\!182$	6.669	2.536	1.154	44.607		
Share of children	58,182	0.446	0.101	0	0.875		

 Table 1: Descriptive Statistics

Notes: This table summarizes statistics for the regression variables. The descriptive statistics for the logged disbursement variables are calculated using only non-zero values, as including 0.01 would result in negative values. Consequently, statistics for logged disbursements are provided exclusively for clusters with positive recorded entries to avoid this issue. All disbursements are reported in U.S. dollars. The raster data on population for certain years and clusters (notably in the Democratic Republic of Congo, Egypt, and Tanzania) indicate zero population values, even though DHS respondents are present. These clusters are in remote regions and additionally, due to the spatial displacement of DHS geolocations by approximately 0-10km where satellite-based population data fail to accurately capture the presence of inhabitants (see Section 5 for robustness checks by buffer sizes).

 $\sum N_{idct}$ is the total number of individuals in that age group in a given cluster. The proportion is a measure of the illiteracy rate among individuals aged 6 to 24 within a specific demographic cluster ⁴. I construct the dependent variable by integrating various self-reported education-related questions, such as educational attainment, schooling status, and other educational variables. For example, in the DHS

⁴The use of enrollment measures as a proxy for educational outcomes has been challenged in the literature, which generally recommends that the outcome variable include educational quality indicators, such as test scores, rather than just enrollment to reveal the true impact (Bennell, 2002). Clemens et al. (2008), for instance, report that higher enrollment rates are associated with declining educational quality in some countries due to factors such as higher student–teacher ratios, increased failure and repetition rates, and lower test scores. However, since disaggregated test score data or other indicators for measuring quality education for a larger sample are unavailable, I use enrollment rates as a proxy for illiteracy in my analysis.

data, some individuals report completing ten years of education but lack entries for schooling status or educational level achieved, and vice versa. Relying solely on one variable, such as educational level, would exclude certain individuals from the sample. To address this, I infer educational levels using other related educational variables where possible (see Appendix, Table A.7). Descriptive statistics are reported in Table 1.

2.2 Main Explanatory Variable: IEG Rating

The World Bank's development projects follow a structured process starting with preparation and moving through implementation to the post-completion evaluation stage. A project is evaluated by the Independent Evaluation Group (IEG) according to a set of criteria (including relevance, efficiency, and effectiveness) and a number of project characteristics. These include whether the World Bank's funding was well designed, how the project was supervised, implementation experience, and the recipient government's initial readiness to implement the project. The overall outcome receives a score from 1 to 6, with 1 denoting highly unsatisfactory and 6 highly satisfactory outcomes (for full detailed discussion on IEG ratings, see Bedasso, 2024; Chauvet et al., 2010; Ika et al., 2012).

The dataset comprises 5,684 World Bank projects approved between 1995 and 2014, valued at over \$630 billion in commitments and \$389 billion in disbursements. My study examines 1,970 geocoded World Bank projects in Africa. The dataset includes project details such as start and end date, sector, and geolocation. In cases where IEG ratings were unavailable, project rating data were matched with project specifics from IEG (2024) for the most recent ratings. The analysis is focused on completed educational projects that have been formally approved and evaluated and have IEG ratings. Fig. 1 illustrates the spatial distribution of DHS survey locations and World Bank aid projects in Africa for only the sampled countries in the analyses. Most World Bank projects are located in East, Central, and West Africa, for which DHS data are available. I define a project as belonging to the educational sector if any of its sub-sector categories (unspecified: education, basic education, secondary education, or post-secondary education) correspond to education.

2.3 Covariates

In compiling the final dataset, I augmented the education and aid data with additional demographic, socioeconomic, and geographic information specific to the cluster location. The selection of these variables was guided by previous research on sub-national aid project analysis (Bitzer & Gören, 2024; Briggs, 2018a, 2018b; Dreher et al., 2008; Nunnenkamp et al., 2016). Following established methodologies in the literature, I systematically considered factors that could influence the relationship between education and



Figure 1: DHS and World Bank Aid Projects Locations in Africa

Notes: The graph displays World Bank projects approved in Africa from 1995 to 2014 (red dots). The green dots show the locations of households surveyed by the DHS program between 1992 and 2020. Brown dots indicate where both aid projects and DHS data are available.

aid effectiveness while excluding variables that might introduce confounding effects. Due to Africa's lack of disaggregated GDP estimates, I used nighttime light images as a proxy for economic development and remoteness. These images have been shown to correlate with official GDP growth data (Doll et al., 2000; Elvidge et al., 1997; Ghosh et al., 2009; Henderson et al., 2012). Utilizing the methodology proposed in the nighttime light literature, I calculated the mean of nighttime light using zonal statistics within each buffer zone. The visible, stable light data were obtained from NASA's Earth Observation Group. They spanned the period 1992 to 2014, with additional recent data from 2015 to 2021 incorporated based on Bitzer and Gören (2024) ⁵. I assumed that aid management is more effective or visible in less remote areas, leading the World Bank to be more likely to allocate aid to regions with higher nighttime light intensity and to be more reluctant to allocate aid to more remote areas.

Furthermore, I included population figures (in hundred thousands) sourced from the Socioeconomic Data and Applications Center (CIESIN, 2018) to mitigate the potential bias stemming from aid targeting densely populated areas (Öhler et al., 2019) and to explore the relationship between population density, educational infrastructure, and aid allocation. Notably, regions with limited educational infrastructure have been found to attract more educational aid from the World Bank (Dollar & Levin, 2006; Song et al., 2020). Next, I include the number of organized conflicts 5 years prior to the survey date from the UCDP Georeferenced Event Dataset (GED), global version 23.1 (Davies et al., 2023; Sundberg & Melander, 2013) in the buffer to proxy for political instability and institutional quality. I assume that conflicts at the location of World Bank projects will delay project approval and execution, will intensify local violence (Blattman & Miguel, 2010; Collier & Hoeffler, 2006; Findley et al., 2023; Maren, 2009; Trisko Darden, 2020; Zürcher, 2019) and may attract more aid when the conflict is over and its effects are more visible (Child, 2014; Wood & Molfino, 2016).

The analyses consider the effect of weather-related variables on household income. Many African regions rely heavily on agriculture as a source of livelihood, making climate variables a significant factor in agricultural productivity (Zamand & Hyder, 2016). Rainfall scarcity or extreme drought due to high temperatures can lead to crop failures, resulting in decreased household incomes. Climate-related economic shocks can cause families to prioritize immediate survival over education, reducing literacy rates as children are compelled to work to support their families instead of attending school (Cockburn & Dostie, 2007; Mahmud & Riley, 2021; Marchetta et al., 2018). Once children drop out of school, it becomes increasingly difficult for them to return (Asare et al., 2023). To isolate this effect, I control for climate variables such as mean precipitation, temperature, and drought. I also control for demographic variables such as average household size, average age, and the share of children (i.e., sons or daughters of the household head) within the cluster as additional factors that correlate with educational outcomes.



Figure 2: Spatial matching approach

Notes: The figure uses Ghana to illustrate the spatial matching approach of World Bank IEG project ratings and DHS locations. A 50 km buffer is drawn around the DHS cluster. The triangles are a subset of the 2019 DHS cluster located in the northern part of Ghana. The squares are overall IEG scores associated with each completed aid project. As shown on the map, Ghana like other countries in my sample does not obtain the two extreme rating scores: 1 (highly unsatisfactory) and 6 (highly satisfactory).

3 Empirical Strategy

3.1 Spatial Matching

I explore the geographic proximity of clusters in the World Bank educational project areas by taking my analysis from the individual level to a more aggregate geographic area (cluster units). I create a 50km buffer, corresponding to an area of about 7,854 km^2 . The buffers are convenient spatial units that allow me to match projects to nearby clusters with educational information. Spatial buffers provide a

 $^{{}^{5}}$ These are calibrated data designed to ensure comparability across different satellites. For detailed explanations of the calibration process, see Bitzer and Gören (2024) and Hsu et al. (2015)

more targeted and precise analysis than administrative regions, making them valuable spatial units for the analysis⁶. The choice of buffer size is important in the estimation as DHS locations do not represent the exact location of the clusters due to geographical displacement of about 0 - 10km (DHS, 2024), to avoid identification of individuals. For instance, a larger buffer size increases the noise by likely counting clusters far from aid projects as beneficiaries of aid, which might not be the case. To circumvent this concern, I provide estimates with different buffer sizes in Section 5. The sample consists of African countries for which DHS survey waves are available. The first wave corresponds to a period with little or no World Bank educational aid activities. It is also worth noting that I examined completed educational aid projects before the cluster was surveyed.

Fig. 2 illustrates the spatial matching approach of aid projects and DHS cluster locations. If an educational aid project was completed before the surveyed date of the DHS cluster, I consider it as treated. I measure aid to education using three approaches: First, an indicator equals one if a cluster has a completed educational aid project. Second, I count the number of completed educational projects. Third, the total value of disbursements⁷ within each cluster is calculated. The disbursement is reported only for the main project. However, these projects often consist of multiple sub-projects across different locations. Therefore, I evenly split the total disbursement among sub-projects under the main project (Briggs, 2018b). All three aid measurement levels must exist before the DHS cluster survey date. I repeat the three steps for the World Bank IEG project rating associated with the project as a proxy for management quality. The overall IEG rating is on a scale from 1 to 6. Therefore, a dummy, a count, and the disbursement value of each specific scale in the buffer are calculated.

Finally, I made 3-point scales to simplify the analysis by combining ratings 1 and 2 as LS projects, 3 and 4 as MS projects, and 5 and 6 as HS projects. The reason is that none of the educational projects completed prior to the cluster survey was rated 6 (highly satisfactory), and only a few clusters were rated

⁶Previous studies in aid have used grid cell analysis to study economic growth/poverty (Bitzer & Gören, 2024; Briggs, 2018b) or conflicts (Wood & Sullivan, 2015). Although a grid cell is an appropriate spatial unit, I am more interested in a more localized effect, where the impact of aid on the illiteracy rate is the primary concern. Grid cells are noisy and broad in area (i.e., often covering larger uniform regions), with no emphasis on areas surrounding an actual DHS cluster (the focus of this analysis). As a result, such a spatial method may include an area that does not contain the outcome variable itself, which may reduce the accuracy of the results.

⁷The aid data on disbursements have limitations, as there are instances where disbursements were not reported for certain years or specific projects. I do not use aid commitments because these are not necessarily the exact amount disbursed and often overstate the actual aid flow. The summary statistics in Appendix table A.1 shows that, on average, approximately \$72 million was disbursed to projects approved during our study period, with approximately \$6.8 million in the educational sector. LS projects on average have the lowest disbursements, amounting to approximately \$380,000, whereas MS projects on average have the highest disbursements.

1 (highly unsatisfactory). Using Ghana as an illustration in Figure 2 (left), there are four MS (three with an IEG rating of 3 [moderately unsatisfactory], and one with an IEG rating of 4 [moderately satisfactory]) and one HS (IEG rating of 5, satisfactory) projects before the survey date of the cluster. Additionally, for each cluster represented by the triangle in Figure 2, I compute the shares and averages for some household characteristics (e.g., average age, average household size) and the outcome variable illiteracy (i.e., the share of individuals between the ages of 6 and 24 years with no formal or completed primary education).

3.2 Identification Strategy

I employ fixed-effects (FE) estimation to investigate whether World Bank management quality (proxied by project rating) has an effect on reducing the illiteracy rate. First, I estimate generic aid impact on illiteracy using Equation (2) and employing the three measurements for World Bank educational aid: an indicator of whether educational aid had ever been completed, a count of the number of completed educational aid projects, and finally, the total disbursement. Second, I explore management quality using Equation (3). The estimated equations take the following form:

$$Illiteracy_{dct} = \beta_1^m Aid_{dct}^m + \mathbf{X}'_{dct} + \delta \mathbf{Z}'_{dct} + \phi \mathbf{G}'_d + \lambda_g + \lambda_t + \varepsilon_{dct}$$
(2)

$$Illiteracy_{dct} = \alpha_1 H S_{dct} + \alpha_2 M S_{dct} + \alpha_3 L S_{dct} + \theta \mathbf{X}'_{dct} + \delta \mathbf{Z}'_{dct} + \phi \mathbf{G}'_d + \lambda_g + \lambda_t + \varepsilon_{dct}$$
(3)

$$Illiteracy_{dct} = \alpha_1 H S_{dct} + \alpha_2 M S_{dct} + \alpha_3 L S_{dct} + \alpha_4 (H S_{dct} \times M S_{dct}) + \alpha_5 (H S_{dct} \times L S_{dct}) + \alpha_6 (M S_{dct} \times L S_{dct}) + \theta \mathbf{X}'_{dct} + \delta \mathbf{Z}'_{dct} + \phi \mathbf{G}'_d + \lambda_g + \lambda_t + \varepsilon_{dct}$$

$$(4)$$

where: *Illiteracy* denotes the share of individuals between 6 and 24 years of age with no education in cluster d in country c at time t, as discussed in Section 2 Equation 1. Aid in Equation (2) captures the impact of World Bank aid on illiteracy across the different aid measurements $m = \{dummy, counts, ln(disbursement)\}$. Equation (3) examines each level of project management quality on illiteracy, and Equation (4) explores interaction effects between management quality⁸. Where: HS = highly satisfactory (scores 5 and 6); MS = moderately satisfactory (scores 4 and 3) and LS = less

⁸For Equation (3) and (4) disbursement of each level of project management quality is log transformed.

satisfactory projects (scores 1 and 2). In Equation (2), β_1^m represents the effect of educational aid on illiteracy for each level of project measurement quality. On the other hand, α_1 to α_6 in Equation (4) captures the marginal impact of project management quality on the illiteracy rate. The marginal effect of a management quality level can be expressed as follows:

$$\frac{\partial Illiteracy_{dct}}{\partial HS_{dct}} = \alpha_1 + \alpha_4 MS_{dct} + \alpha_5 LS_{dct} \tag{5}$$

$$\frac{\partial Illiteracy_{dct}}{\partial MS_{dct}} = \alpha_2 + \alpha_4 HS_{dct} + \alpha_6 LS_{dct} \tag{6}$$

$$\frac{\partial Illiteracy_{dct}}{\partial LS_{dct}} = \alpha_3 + \alpha_5 HS_{dct} + \alpha_6 MS_{dct} \tag{7}$$

In the parsimonious specification in Equation (3), the effect of the main management quality level on illiteracy is straightforward and reflects its unique influence. However, when interaction terms are included, the interpretation changes significantly. In such cases, the impact of management quality depends on the levels of the interacted variables, which can lead to coefficients of the interacted variables having the opposite signs from the specifications without interaction terms. That is, the effect is no longer isolated but contingent on the interplay between variables. The vector \mathbf{X}' socioeconomic controls (e.g., number of conflicts, population in hundred thousands and mean nighttime light). The vector \mathbf{Z}' is climatic controls (mean temperature, precipitation, and drought), and \mathbf{G}' is a set of DHS cluster demographics (average age, average household size, and share of children). λ_g^9 and λ_t are administration level 2 and year FE, respectively.

Administration level 2 and year FE control for unobserved variations that are constant over time and vary across clusters. This includes differences in humanitarian needs or other pertinent conditions (historical and cultural relations) that influence the allocation of aid from the World Bank and other bilateral sources. This ensures that I am always comparing within a given cluster at a single point in time. Finally, ε is an idiosyncratic error term. Unless otherwise stated, standard errors are clustered at DHS cluster level d throughout all regressions.

4 Main Results

The baseline results are divided into three sections based on the measurement of educational aid, namely: indicator, count, and disbursement measures. Each of the reported sections is further structured into

⁹I replace countries with no global administrative area level 2 (gadm) with level 0 to avoid them being dropped out of the analysis. Excluding them also does not alter the significance or sign of the coefficients. Results are available upon request.

three columns for all regression tables. Column (1) of Tables 2, 4, 6 provides estimates for Equation (2) examining the generic impact of completed World Bank educational aid projects on illiteracy. Column (2) focuses on World Bank project management quality using Equation (3) by estimating a parsimonious model without the interaction terms to investigate the influence of management quality on illiteracy. Finally, Column (3) uses Equation (4) to explore the interaction effects between management quality levels to ascertain whether illiteracy rates in one area are influenced by the management quality of other educational projects in the same vicinity. In addition, to aid interpretation, the dependent variable, the share of illiteracy (proportion of individuals aged 6 to 24 who lack formal education), was scaled by a factor of 100 to express it as a percentage points. Consequently, the coefficients of the explanatory variables reflect the change in the illiteracy share in percentage points. Standard errors are clustered at the DHS cluster level. For simplicity, coefficients for climate variables and DHS demographics are controlled but not reported.

Indicator for Educational Aid: Table 2 investigates the impact of completed educational aid projects on illiteracy rates, employing an indicator for the presence of educational aid before the cluster survey date. In Column (1), the analysis reveals a positive and statistically significant coefficient at the 1% level. This indicates that, all else being equal, starting a new educational project in a specific cluster and year is associated with an increase in illiteracy by approximately 1.75 percentage points. This result is unexpected, as educational aid is typically aimed at improving literacy rather than increasing illiteracy. One potential explanation is that some projects may have been poorly implemented or unsuccessful, resulting in little to no improvement in literacy rates. This raises concerns about whether World Bank educational aid is effectively achieving its goal of reducing illiteracy. A limitation of this specification is the use of a broad indicator for educational aid presence, which may overlook important factors such as the role of management quality in shaping project outcomes. Incorporating measures of management quality, such as overall performance ratings from the IEG, could provide valuable insights into these puzzling findings. These considerations are addressed in subsequent models to better understand the underlying factors driving these results.

The analysis further reveals that conflicts, which are significant at 1%, play a significant role in increasing illiteracy rates. The disruptions to normal human activities such as schooling, can result in schools closing and children being unable to attend classes. This finding highlights the urgent need to address conflict situations to ensure the continuity of education for all. Similarly, rapid population growth though insignificant in Column (1) but positive and significant at 5% significance level in the remaining columns is associated with higher illiteracy rates. Particularly in regions like Africa, where educational infrastructure is often insufficient to accommodate a large number of students. Overcrowded classrooms

Table 2: Indicator: If Cluster Received any Educational Aid								
	(1)	(2)	(3)					
		iable: Illiteracy						
Variables	Educational Aid	Rated Projects	Interactions					
Educational project	1.753***							
110	(0.3014)							
HS		-2.634***	-0.2665					
		(0.4395)	(0.5406)					
MS		0.1794	ì.590***					
TO		(0.3138)	(0.3703)					
LS		À.718***	8.152***					
		(0.5704)	(0.8353)					
HS $ imes$ MS			-4.006***					
			(0.5796)					
$\mathrm{HS} imes \mathrm{LS}$			-4.003***					
			(0.9465)					
$MS \times Low$			-3.382***					
			(0.6681)					
Num. conflicts 5 years prior	0.0183***	0.0185***	0.0153* [*]					
	(0.0069)	(0.0069)	(0.0069)					
Population per 100,000	0.0169	0.0323**	0.0321* [*]					
	(0.0133)	(0.0132)	(0.0132)					
Year	Yes	Yes	Yes					
District FE	Yes	Yes	Yes					
Weather Controls	Yes	Yes	Yes					
Cluster demographics	Yes	Yes	Yes					
Clusters with some lady aid	91.676	91.676	91 676					
Clusters with compl. edu. aid	21,676	$21,\!676 \\ 37,\!924$	21,676 27,024					
Clusters with compl. aid Clusters with edu. aid	$37,924 \\ 37,169$	37,924 37,169	37,924 37,160					
Clusters with aid	$57,109 \\ 57,341$	57,109 57,341	$37,169 \\ 57,341$					
Num. DHS clusters	57,341 57,341	57,341 57,341	57,341 57,341					
Observations	57,541 58,182	57,541 58,182	57,541 58,182					
Within \mathbb{R}^2	0.01100	0.01196	0.01378					
	0.01100	0.01190	0.01370					

Notes: The dependent variable is the share of individuals between the ages of 6 and 24 who are illiterate (see Section 2 for definition of illiteracy) multiplied by 100. Each column comes from a unique regression. Weather-related controls include mean precipitation, mean temperature, and mean drought. DHS cluster controls include the average age in the cluster, average household size, and the share of sons and daughters in the cluster. All regressions include year and district (that is, administrative level 2) FE. Clustered standard errors by DHS clusters in parentheses. Signif. Codes: ***: 0.01, **: 0.05, *: 0.1

and the inability of school systems to absorb the growing population exacerbate the problem, leaving many children without access to education (UNESCO, 2024b).

Column (2) investigates further to differentiate projects based on overall managerial quality using the IEG rating obtained. In contrast to earlier estimates in Column (1), I found negative and statistically

Aid Management and Educational Outcome

Table 3: Marginal Effects – Indicator							
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
HS	MS	\mathbf{LS}	Freq.	$\operatorname{Freq.}(\%)$	Μ	arginal Effec	ets
					HS	MS	LS
0	1	0	$7,\!645$	38.82	_	1.59	_
1	1	0	$4,\!484$	22.77	-4.01	-2.42	_
1	0	0	4,051	20.57	0	_	_
0	1	1	1,061	5.39	—	-1.79	4.77
1	1	1	886	4.50	-8.01	-5.80	0.77
0	0	1	620	3.15	_	_	8.15
1	0	1	609	3.09	-4.00	—	4.15
0	2	0	191	0.97	—	1.59	_
0	2	2	54	0.27	—	-5.17	1.39
2	0	0	39	0.20	0	—	_
2	2	0	38	0.19	-8.01	-6.42	_
2	0	2	10	0.05	-8.01	—	0.15
0	0	2	4	0.02	_	—	8.15
0	3	3	2	0.01	_	-8.56	-1.99
0	3	0	1	0.01	—	1.59	—

Notes: This table quantifies the marginal effects of combinations of managerial quality types (HS, MS, LS) on the illiteracy rate. Columns (1)–(3) represent unique combinations of project quality types observed in the data, while columns (4)–(5) provide the frequency of each combination as raw counts and percentages. Marginal effects in columns (6)–(8) are computed using Equations (5)–(7) and coefficients from Table 2, column (3). For example, the marginal effect for a single MS project (row 1) is derived using Equation (6): $1.590-4.006 \times HS_{dct} - 3.382 \times LS_{dct}$. Substituting $HS_{dct} = 0$ and $LS_{dct} = 0$, the result is -1.59 percentage points. Insignificant coefficients are treated as zero. Calculations are performed with maximum decimal and rounded to two decimal places. Bolded values are mentioned in the paper.

significant effect at 1% for HS educational project. Implying that when HS educational project exist in an area, illiteracy is reduced by approximately 2.63 percentage points. While I do not find significant effects for MS project. HS projects are likely to be more effective in improving education. However, LS project is associated with higher illiteracy rates of about 4.72 percentage points. It suggests that poorly managed projects fail to achieve their intended goals, leading to a continued growth in the illiteracy rate in a given spatial unit. Poor management can lead to inefficiencies and resource wastage in addressing the needs of the target population. This suggests that examining the presence of educational aid without considering other factors such as the quality of project management can be misleading.

Table 2, Column (3), evaluates the dependencies of aid management quality, specifically analyzing whether the coexistence of different educational management quality projects within the same cluster can improve illiteracy. The corresponding marginal effects for unique combinations of project quality categories are presented in Table 3. For example, the presence of a single MS project, which is the most frequently observed case in the data (38.82%), results in a significant positive marginal effect, increasing illiteracy rates by approximately 1.59 percentage points (Table 3, row 1; see table footnotes for calculation

16

	(1) Dependent V	(2) ariable: Illiteracy (6	(3)
Variables	Educational Aid	Management Rating	Interactions
Num. educational projects	0.0877***		
HS	(0.0271)	-0.7783***	-0.4049***
MS		$egin{array}{c} (0.1285) \ 0.3194^{***} \end{array}$	$egin{pmatrix} (0.1453) \ 0.5228^{***} \ \end{bmatrix}$
LS		$egin{array}{c} (0.0810) \ 2.193^{***} \end{array}$	(0.0938) 2.769^{***}
$HS \times MS$		(0.2505)	(0.3050) - 0.2635^{***}
$HS \times LS$			(0.0508) - 0.3952^{**}
$\mathrm{MS} imes \mathrm{LS}$			(0.1833) - 0.1231^{***}
Num. conflicts 5 years prior	0.0186***	0.0196***	$egin{array}{c} (0.0428) \ 0.0157^{**} \end{array}$
Population per 100,000	$(0.0068) \\ 0.0232^* \\ (0.0134)$	$egin{array}{c} (0.0069) \ 0.0260^{**} \ (0.0133) \end{array}$	$egin{array}{c} (0.0070) \ 0.0248^{*} \ (0.0133) \end{array}$
Year	Yes	Yes	Yes
District FE Weather controls	Yes Yes	Yes Yes	Yes Yes
Cluster demographics	Yes	Yes	Yes
Clusters with compl. edu. aid Clusters with compl. aid Clusters with edu. aid Clusters with aid Num. DHS clusters Observations	$21,676 \\ 37,924 \\ 37,169 \\ 57,341 \\ 57,341 \\ 58,182$	21,676 37,924 37,169 57,341 57,341 58,182	$21,676 \\ 37,924 \\ 37,169 \\ 57,341 \\ 57,341 \\ 58,182$
Within R ²	0.01038	0.01207	0.01259

 Table 4: Counts: Number of Completed Educational Projects

Notes: The dependent variable is the share of individuals between the ages of 6 and 24 years who are illiterate (see Section 2 for definition of illiteracy) multiplied by 100. Each column comes from a unique regression. Weather-related controls include mean precipitation, mean temperature, and mean drought. DHS cluster controls include the average age in the cluster, average household size, and the share of sons and daughters in the cluster. All regressions include year and district (that is, administrative level 2) FE. Clustered standard errors by DHS clusters in parentheses. Signif. Codes: ***: 0.01, **: 0.05, *: 0.1

details). Similarly, a single LS project, in the absence of other projects, leads to an even greater increase in illiteracy, approximately 8.15 percentage points (row 6 column (8)). Conversely, no significant effect is observed for clusters containing only HS projects (Table 3, row 3), which represent the third-highest frequency in the data.

In cases where at least one HS and one MS project coexist within a cluster, the marginal effects indicate reductions in illiteracy rates by 4.01 and 2.42 percentage points, respectively (Table 3, row 2).

These results suggest that clusters with both HS and MS projects experience a significant decline in the proportion of uneducated individuals aged 6–24 years. In contrast, clusters with a combination of HS and LS projects show opposing effects, with marginal reductions of approximately 4 percentage points for HS and increases of 4.15 percentage points for LS. In conclusion, the findings in Table 3 indicate that a combination of HS and MS projects leads to improved literacy rates, whereas clusters with only LS projects exhibit adverse outcomes. These findings highlight the critical importance of synergy between different types of educational aid projects, to achieve meaningful positive impacts on educational outcomes within a given spatial unit.

Counts of Educational Aid: Table 4 presents the results for completed educational aid by counting the number of projects per cluster. Column (1) results indicate that the mere number of completed educational projects does not reflect an improvement in literacy. This finding suggests that providing more educational resources might not necessarily produce the desired outcome. Probing further into project performance in Column (2) also confirms the results in Table 2 Column (2). Indicating that an additional completed HS project results in a reduction in the illiteracy rate of 0.78 percentage points. This finding emphasizes the critical role of not just the quantity but also the quality of educational projects in improving outcomes. The results further suggest that increasing the number of projects is insufficient to reduce illiteracy if those projects are not effectively managed. Effective management is essential to ensure that the resources and efforts invested in educational initiatives translate into meaningful and tangible results. That is, the combination of these projects with robust management practices will drive the rapid achievement of SDG 4. An additional implemented MS and LS project, in contrast, increase illiteracy by about 0.32 and 2.19 percentage points, respectively, signaling a more damaging effect on educational outcome for LS educational projects.

Column (3) examines the effects of interactions between different types of projects. Table 5 presents the marginal effects for a subset of unique project quality combinations observed in the data, focusing on combinations with a percentage frequency of $\geq 1\%$ (see table footnotes for details on the calculation of marginal effects)¹⁰. The results in Table 5 indicate that HS projects consistently reduce illiteracy rates. For example, areas with only HS projects, which occur 11.6% of the time, experience a marginal effect of approximately -0.40 percentage points (rows 2, 5, 14, and 17). Conversely, MS projects typically increase illiteracy rates, except when coexisting with HS projects in instances where HS projects outnumber MS projects. In such cases, the marginal effects of MS projects can turn negative. For instance, in the last row of Table 5, the marginal effect of MS becomes negative when at least six HS projects and one MS

¹⁰Marginal effects for remaining project combinations are available upon request.

Aid Management and Educational Outcome

	Table 5: Marginal Effects – Counts								
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)		
HS	MS	LS	Freq.	$\operatorname{Freq.}(\%)$	Μ	arginal Effec	ts		
					HS	MS	LS		
0	1	0	3,919	19.90	_	0.52	_		
1	0	0	2,284	11.60	-0.40	—	_		
0	2	0	1,859	9.44	—	0.52	_		
1	1	0	$1,\!139$	5.78	-0.67	0.26	_		
2	0	0	733	3.72	-0.40	_	_		
0	3	0	713	3.62	—	0.52	_		
0	1	1	530	2.69	—	0.40	2.65		
0	0	1	487	2.47	—	—	2.77		
2	1	0	480	2.44	-0.67	-0.00	_		
0	6	0	466	2.37	_	0.52	_		
2	2	0	404	2.05	-0.93	-0.00	_		
1	3	0	386	1.96	-1.20	0.26	_		
0	4	0	385	1.95	_	0.52	_		
3	0	0	330	1.68	-0.40	_	_		
0	5	0	291	1.48	_	0.52	_		
1	2	0	255	1.29	-0.93	0.26	_		
4	0	0	245	1.24	-0.40	_	_		
2	0	1	227	1.15	-0.80	_	1.98		
6	1	0	202	1.03	-0.67	-1.06	_		

 Table 5: Marginal Effects – Counts

Notes: This report analyzes marginal effects for project combinations appearing with $\geq 1\%$ frequency (222 unique combinations) for project counts. Columns (1)–(3) present unique project quality combinations, whereas columns (4)–(5) show raw counts and percentages. Columns (6)–(8) display marginal effects calculated using Equations (5)–(7) and coefficients from Table 4, column (3). For example, for a single HS project, substituting $MS_{dct} = 0$ and $LS_{dct} = 0$ into $-0.4049 - 0.2635 \times MS_{dct} - 0.3952 \times LS_{dct}$ yields -0.40 percentage points. Insignificant coefficients are treated as zero. Bolded values are mentioned in the paper. Calculations use maximum decimals and are rounded to two decimal places.

project coexist within the same geographical area.

LS projects, on the other hand, consistently worsen illiteracy. A single LS project, which occurs in 2.47% of the data, leads to a 2.77 percentage point increase in illiteracy rates. The marginal effect of LS projects only turns negative in rare cases where HS projects significantly outnumber LS projects. For example, when at least eight HS projects and two LS projects coexist, the marginal effect becomes -0.39 percentage points (not shown in Table 5). However, such cases are exceptionally rare, occurring in only 0.005% of the data¹¹. In summary, the findings in Table 5 suggest that HS projects are the only project management category that consistently improves literacy rates. By contrast, MS and LS projects, either independently or in combination, generally have the opposite effect. These results reinforce the importance

¹¹Table 5 reports only combinations with a frequency of $\geq 1\%$, emphasizing the rare occasions where LS projects exhibit negative effects. Such cases typically occur when LS coexist with a substantial number of HS projects.

of considering interactions between different project categories, as their coexistence can produce varying effects on educational outcomes within an area.

	(1)	(2)	(3)
		ariable: Illiteracy (6	
Variables	Educational Aid	Management Rating	
ln(Value Edu.)	0.0846***		
$\ln(\mathrm{HS})$	(0.0155)	-0.1329^{***}	-0.1089^{***}
$\ln(MS)$		(0.0216) 0.0139 (0.0156)	(0.0233) 0.0041 (0.0160)
$\ln(LS)$		(0.0156) 0.2428^{***}	(0.0160) 0.3389^{***}
$\ln(\text{HS} \times \text{MS})$		(0.0305)	(0.0351) -0.0091*** (0.0014)
$\ln(MS \times LS)$			(0.0014) - 0.0102^{***}
$\ln(MS \times LS)$			(0.0024) - 0.0080^{***}
Num. conflicts 5 years prior	0.0184***	0.0186***	(0.0017) 0.0154^{**}
Population per 100,000	$(0.0068) \\ 0.0163 \\ (0.0133)$	$egin{array}{c} (0.0069) \ 0.0315^{**} \ (0.0132) \end{array}$	$egin{array}{c} (0.0069) \ 0.0315^{**} \ (0.0132) \end{array}$
Year	Yes	Yes	Yes
District FE	Yes	Yes	Yes
Weather controls	Yes	Yes	Yes
Cluster demographics	Yes	Yes	Yes
Clusters with compl. edu. aid	21,676	21,676	21,676
Clusters with compl. aid Clusters with edu. aid	$37,924 \\ 37,169$	$37,\!924 \\ 37,\!169$	$37,924 \\ 37,169$
Clusters with aid	57,341	57,341	57,341
Num. DHS clusters	57,341	57,341	57,341
Observations	58,182	58,182	58,182
Within \mathbb{R}^2	0.01092	0.01192	0.01374

Table 6: Disbursement	: Total Value	e of Educational Aid
-------------------------------	---------------	----------------------

Notes: The dependent variable is the share of individuals between the ages of 6 and 24 years who are illiterate (see Section 2 for definition of illiteracy) multiplied by 100. All disbursements are logged transformed using $\ln(0.01+..)$. Each column comes from a unique regression. Weather-related controls include mean precipitation, mean temperature, and mean drought. DHS cluster controls include the average age in the cluster, average household size, and the share of sons and daughters in the cluster. All regressions include year and district (that is, administrative level 2) FE. Clustered standard errors by DHS clusters in parentheses. Signif. Codes: ***: 0.01, **: 0.05, *: 0.1

Disbursement of Educational Aid: In Table 6, I used the log of total disbursements to measure educational aid. As in previous results, Column (1) shows a positive effect, which implies that a 1%

dollar increase in disbursement leads to approximately 0.0008% increase in illiteracy. That is, the volume of educational aid does not necessarily improve illiteracy, which aligns with the results with indicators and count measures. In Column (2), an increase in disbursement for HS projects significantly decreases the illiteracy rate by about 0.001%. At the mean, it leads to a reduction of about 2.1%. In Column (3), the total marginal effect for an increase in disbursements for HS projects consistently leads to a significant reduction in illiteracy ($-0.1089 - 0.0091 \times ln(MS_{dct}) - 0.0102 \times ln(LS_{dct})$). At the mean, this is approximately -0.39 ($-0.1089 - 0.0091 \times 15.4 - 0.0102 \times 14.3$; for mean values¹², see Table 1). For MS and LS projects, evaluating at the mean will be approximately -0.26 ($-0.0091 \times ln(HS_{dct}) - 0.0080 \times ln(MS_{dct}) \Rightarrow 0.0389 - 0.0102 \times 15.9 - 0.0080 \times 14.3$) and 0.05 ($0.3389 - 0.0102 \times ln(HS_{dct}) - 0.0080 \times ln(MS_{dct}) \Rightarrow 0.3389 - 0.0102 \times 15.9 - 0.0080 \times 15.4$), respectively, on illiteracy rate. The impact of educational aid on illiteracy diminishes with project management quality.

5 Robustness Checks

The primary findings are examined to address two main concerns. First, I assess the sensitivity of the baseline results to two alternative sample restrictions—urban and rural—to determine whether educational aid and management effectiveness on illiteracy rates differ. Second, I show results for different buffer sizes. Table 7 presents the urban–rural sample results for the three baseline estimates.

In analyzing the indicator measures for both educational aid and management quality in Panels A and B, a similar positive effect (1.11 and 1.88 percentage increase for urban and rural areas, respectively) of completed projects on illiteracy rates is observed. However, when considering project management, additional HS projects in urban clusters improve literacy by approximately 3.76 percentage points (column (2)), whereas MS and LS management quality show no or some positive significant effects, respectively, for additional implemented educational projects. In contrast, an HS project in a rural cluster results in an estimated 1.96 percentage point reduction in illiteracy rates (column (5)), whereas an LS project results in an estimated increase in illiteracy (urban = 5.53; rural = 4.15 percentage points). The key difference between urban and rural areas is most evident in the stronger impact on illiteracy in urban compared to rural areas.

Panels A and B, column (3), report the coefficients of interactions between project management quality categories for the indicator measure. The marginal effects for urban and rural clusters with respect to the indicator measure are detailed in Appendix Tables A.2 and A.3 Panel A, respectively, with an example of calculation provided in the table notes. In urban areas, 44% of educational projects are classified as

¹²Note that the reported means do not include clusters with zero disbursements.

	Table 7	: Replication of Ba	aseline Results:	: Sample Split		
	(1)	(2)	(3)	(4)	(5)	(6)
Variables	T 1			Illiteracy (6-24 y		T
	Educational Aid	Management rating A: Indicator – U			Management Rating el B: Indicator-Rur	
Educational project	1.111***	n. matcator c	roun	1.875***	a D. Matcalor Rur	<i>ui</i>
	(0.3910)			(0.3895)		
HS		-3.762^{***}	-1.027		-1.964^{***}	0.1252
MS		$(0.6798) \\ -0.3108$	$(0.8998) \\ 0.9750^{**}$		$(0.5367) \\ 0.0389$	(0.6572) 1.548^{***}
NID .		(0.4143)	(0.4599)		(0.4014)	(0.4822)
LS		5.526^{***}	8.246^{***}		4.152^{***}	9.213^{***}
		(1.054)	(1.383)		(0.6794)	(1.049)
$HS \times MS$			-4.154^{***}			-3.600^{***}
$HS \times LS$			(0.9244) -2.743*			(0.7207) -5.828***
			(1.458)			(1.239)
$MS \times LS$			-3.178***			-4.951***
			(1.013)			(0.9041)
Within \mathbb{R}^2	0.02234	0.02539	0.02833	0.00832	0.00871	0.01094
	Pane	l C: Counts – Ur	ban	Pan	nel D: Counts-Rura	l
Num. educational projects	-0.0608**			0.1447***		
HS	(0.0304)	-1.217***	-0.7080***	(0.0489)	-0.6318***	-0.3317^{*}
115		(0.2174)	(0.2424)		(0.1560)	(0.1773)
MS		0.1363	0.3780***		0.1216	0.3442**
		(0.1011)	(0.1141)		(0.1314)	(0.1519)
LS		1.513***	1.844***		3.219^{***}	4.332***
$HS \times MS$		(0.3166)	(0.3781) - 0.4169^{***}		(0.4270)	(0.5189) - 0.1660^{**}
IIS × MS			(0.0804)			(0.0659)
$HS \times LS$			0.0706			-0.7359***
			(0.2821)			(0.2405)
$MS \times LS$			-0.1337***			-0.3125***
			(0.0457)			(0.1132)
Within R ²	0.02204	0.02506	0.02694	0.00770	0.00909	0.00962
		: Disbursement –	Urban		F: Disbursement-R	ural
ln(value edu.)	0.0624***			0.0809***		
$\ln(HS)$	(0.0200)	-0.1784***	-0.1227***	(0.0203)	-0.1023***	-0.1120***
		(0.0329)	(0.0350)		(0.0267)	(0.0300)
$\ln(MS)$		-0.0195	-0.0334		0.0063	-0.0149
		(0.0203)	(0.0208)		(0.0203)	(0.0213)
$\ln(LS)$		0.2757^{***}	0.3403^{***}		0.2100^{***}	0.3616^{***}
$\ln(\text{HS} \times \text{MS})$		(0.0549)	(0.0602) - 0.0089^{***}		(0.0370)	(0.0439) -0.0081***
$m(115 \times m5)$			(0.0021)			(0.0018)
$\ln(\text{HS} \times \text{LS})$			-0.0061*			-0.0151***
			(0.0035)			(0.0033)
$\ln(MS \times LS)$			-0.0073***			-0.0129***
			(0.0026)			(0.0024)
Within R ²	0.02246	0.02524	0.02790	0.00807	0.00864	0.01097
Clusters with compl. edu. aid	8,159	8,159	8,159	13,517	13,517	13,517
Clusters with compl. aid	14,202	14,202	14,202	23,722	23,722	23,722
Observations	21,205	21,205	21,205	36,977	36,977	36,977
Year FE	Yes	Yes	Yes	Yes	Yes	Yes
District FE Baseline controls	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes
Dasenne controis	168	res	ies	res	IES	168

 Table 7: Replication of Baseline Results: Sample Split

Notes: This table shows sample split for urban and rural clusters. The dependent variable is the share of individuals between the ages of 6 and 24 years who are illiterate (see Section 2 for definition of illiteracy) multiplied by 100. All disbursements are logged transformed using $\ln(0.01+..)$. Each column comes from a unique regression. Baseline controls include: geographic controls (number of conflicts 5 years prior to the survey date, population per 100,000 and mean nighttime light); weather (mean precipitation, mean temperature and mean drought) and DHS cluster controls (average age in the cluster, average household size and the share of sons and daughters) in the cluster. All regressions include year and district (that is, administrative level 2) FE. Clustered standard errors by DHS clusters in parentheses. Signif. Codes: ***: 0.01, **: 0.05, *: 0.1

MS. These projects exhibit a positive marginal impact on illiteracy, with an effect size of approximately 0.98 percentage points (Appendix Table A.2, Panel A, column (7)). By contrast, in rural areas, MS projects (comprising the largest share at 35.85%) have a stronger negative effect, increasing illiteracy by approximately 1.55 percentage points. HS projects consistently reduce illiteracy in both urban and rural

areas, regardless of other coexisting educational projects. However, HS projects alone show no significant marginal effect. For example, in cases where two HS and LS projects coexist within the same vicinity (observed only 4 times in the urban sample and 6 times in the rural sample) the marginal effect of HS is -5.49 percent points in urban areas and -11.66 percent points in rural areas.

Interestingly, whereas MS projects alone contribute to higher illiteracy, their coexistence with other project types (either HS or LS) results in negative marginal effects in both urban and rural communities. In urban settings, the marginal effect of at least one MS and one HS project is a 3.18 percent unit decrease in illiteracy, whereas in rural areas, the effect is -2.05 percentage points. Similarly, when an MS project coexists with one LS project, the marginal effect of MS is -2.20 percentage points in urban areas and -3.40 percentage points in rural areas. Another notable finding is the variation in effect size when an MS project coexists solely with HS or LS projects. In urban areas, the effect size for MS coexisting with HS is much larger than in rural areas. However, the reverse is true when MS coexists only with LS projects. Finally, LS projects exhibit a negative marginal effect in rural areas when coexisting with more than one other project category (HS or MS). For example, in rural areas, when LS projects share the same buffer zone with two MS or HS projects, the marginal effects are -0.69 and -2.44 percentage points, respectively. Such effects are not observed in urban areas. Similarly, when all three project categories (HS, MS, LS) coexist, the marginal effect for LS in rural areas is -1.57 percentage points, while in urban areas, it is positive at 2.33 percentage points, holding other factors constant.

Panels C and D provide estimates for the number of completed projects and their effects on illiteracy rates. While urban clusters experience a reduction of 0.06 percentage points in illiteracy with additional completed projects (Panel C, column (1)), rural communities show a contrasting increase of 0.14 percentage points (Panel D, column (4)) holding all things constant. However, a further analysis of project management quality offers a more nuanced understanding. In rural areas, the total number of educational projects alone appears to increase illiteracy rates, but the presence of HS projects significantly improves outcomes. Specifically, HS projects reduce illiteracy by approximately 0.63 percentage points in rural areas (Panel D, column (5)) and by an even greater magnitude of 1.2 percentage points in urban areas (Panel C, column (2)). In contrast, MS and LS projects show either insignificant or positive effects on illiteracy, respectively.

Examining the marginal effects (see Appendix Tables A.2 and A.3, Panel B and table notes for calculations) of interactions between management quality categories reveals consistent reductions in illiteracy rates for HS projects, whether alone or coexisting with other MS and LS projects. For example, HS projects alone lead to a reduction in illiteracy of 0.71 percentage points in urban areas and 0.33 percentage points in rural areas. MS projects, when they exist alone, worsen illiteracy by 0.38 percentage points in urban areas and 0.34 percentage points in rural areas. Interestingly, in urban areas, the marginal effect of MS becomes negative when coexisting with one or more HS projects. For instance, with two HS projects and one MS project, the marginal effect for MS is -0.46 percentage points in urban areas but remains negligible at 0.01 percentage points in rural areas. LS projects consistently show positive marginal effects, exacerbating illiteracy rates even when coexisting with other projects, but seldom coexist with MS or HS project categories.

Finally, using the disbursement measure in Panels E and F of Table 7, we observe similar patterns for both urban and rural areas. For project interactions (Columns (3) and (6)), the marginal effect of an HS project, when evaluated at the mean, reduces the illiteracy rate by approximately 0.35 percentage points in urban areas $(-0.1227 - 0.0089 \times MS - 0.0061 \times LS \Rightarrow -0.1227 - 0.0089 \times 15.4 - 0.0061 \times 14.3 = -0.35$; see Table 1 for mean values) and by 0.45 in rural areas $(-0.1120 - 0.0081 \times MS - 0.0151 \times LS \Rightarrow -0.1120 - 0.0081 \times 15.4 - 0.0151 \times 14.3 = -0.45)$ ceteris paribus. An MS project reduces illiteracy by approximately 0.25 percentage points in urban and 0.31 in rural clusters (Urban: $-0.0089 \times ln(HS_{dct}) - 0.0073 \times ln(LS_{dct}) \Rightarrow -0.0089 \times 15.9 - 0.0073 \times 14.3 \approx -0.25$; Rural: $-0.0081 \times ln(HS_{dct}) - 0.0129 \times ln(LS_{dct}) \Rightarrow -0.0081 \times 15.9 - 0.0129 \times 14.3 \approx -0.31$). An LS project, on the other hand, shows an increase in illiteracy of 0.13 percentage points in urban areas $(0.3403 - 0.0061 \times ln(HS_{dct}) - 0.0073 \times ln(MS_{dct}) \Rightarrow 0.3403 - 0.0061 \times 15.9 - 0.0073 \times 15.4 \approx 0.13$), but a reduction in illiteracy by 0.08 percentage points in rural areas, all else being equal $(0.3616 - 0.0151 \times ln(HS_{dct}) - 0.0129 \times ln(MS_{dct}) \Rightarrow 0.3616 - 0.0151 \times 15.9 - 0.0129 \times 15.4 \approx -0.08$).

From Table 7, it can be deduced that there are small, significant differences between urban and rural clusters for the indicator and count analyses. The effects observed in the main analysis are consistent across both contexts—any positive changes associated with educational aid yield equivalent results in both urban and rural communities, but the magnitude of change is higher in urban than in rural areas. For disbursement, rural areas show a stronger reduction in illiteracy for the HS and MS projects (HS = -0.45% and MS = -0.31%) compared to urban clusters (HS = -0.35%, MS = -0.25%). Although LS projects reduce illiteracy in rural areas (-0.08%), they increase illiteracy in urban communities (0.13%). The results suggest that project funding can contribute immensely to reducing rural illiteracy when managed well. If the educational aid is effectively managed, it can produce a slightly higher positive outcome in both rural and urban settings. This suggests that the quality of World Bank educational project management is a more important factor than the geographic location, volume of funding, or number of projects.

Finally, I investigate whether these findings hold when using different buffer sizes, in case there is selection bias. Specifically, I consider whether a 5 km buffer interval (i.e., 20, 25,..., 45 km) might bias the results. Tables A.4, A.5 and A.6 demonstrate that the main results are not sensitive to the choice

of buffer size. The results are essentially similar to the baseline findings, with only minor changes in significance in specific columns. For instance, in Panels B to F of Table A.4 Column (3), the HS project is significant at 5% and 1% and in Table A.6 Panels C and E Column (3), the interaction term $HS \times LS$ and $ln(HS \times LS)$ is not significant compared to the baseline results in Tables 4 and 6, respectively. This means that changes in buffer size do not significantly alter the observed effects. The measurements used to evaluate the impact of educational aid are precise enough to capture the effects consistently, regardless of slight variations in the buffer size.

6 Discussion

This study's empirical analysis has shown a heterogeneous relationship between the quality of management in World Bank–funded educational projects and illiteracy rates in more than 57,000 DHS clusters in Africa. The picture that emerges for aggregate educational aid is consistent. The three measures—indicator, count and disbursement volume of completed educational aid projects—show a significant rise in illiteracy. The extended analyses on management quality indicate that highly satisfactory projects improve literacy rates, in contrast to moderately satisfactory and less satisfactory projects and the findings remain robust across measures and buffer sizes. Additionally, I found interaction effects between different categories of management quality, showcasing that management quality of projects influence the effectiveness of other projects within a shared spatial unit. An HS project, for example, will positively influence LS projects within the same geographical unit. The findings carry significant implications for our understanding of the quality and effectiveness of educational aid management.

To begin, my analyses of educational aid impact on illiteracy rates revealed statistically significant increases in illiteracy ranging from 0.8 to 1.8 percentage points. Without further investigation into the quality of management in these projects, one might conclude that educational aid does not improve educational outcomes. Sole reliance on an indicator of the presence or absence of educational aid may not adequately reflect the effectiveness of project execution, which may subsequently impact the project's intended outcomes. Therefore, in evaluating educational aid's impact on specific outcomes such as illiteracy, there must be a measure that considers management quality, overall project performance and the intensity or scale of the project. Disregarding these may affect conclusions drawn. For example, Yogo (2017) emphasized that achieving the goal of universal primary education would necessitate more than doubling the current levels of aid. Yet this projection might change if existing projects were more effectively managed and implemented. The findings in Table 6, column 1, suggest that an additional dollar of aid increases illiteracy. When management quality is considered (as shown in column 2 of Table 6), however, it becomes evident that it is poor management that significantly undermines project outcomes. Each additional dollar allocated to HS projects reduces illiteracy by approximately 0.001 percentage points, translating into an average reduction of 2.1 percentage points, ceteris paribus. Conversely, MS and LS projects exhibit opposite effects, further emphasizing the need to enhance managerial oversight rather than merely increase aid volumes. Since World Bank educational projects are significant developmental initiatives designed to offer incentives and resources to governments grappling with complex economic challenges that impede growth (Stubbs et al., 2016), the overall performance of project management must be considered when measuring aid effectiveness.

Crucially, my analysis demonstrates that the quality of management oversight in educational aid can have a significant positive or negative effect on the overall impact of a project. For instance, the findings indicate that highly satisfactory educational projects consistently lead to reductions in illiteracy rates, whereas moderately and less satisfactory projects fail to achieve such outcomes. Effective management not only minimizes misappropriation of project funds but also reduces errors at critical stages of the project lifecycle, such as approval, implementation, supervision and monitoring, which are all essential for achieving the intended objectives (Ika et al., 2012; Kilby, 2000). A persistent challenge highlighted in the literature, particularly in Africa, is the prevalence of corruption and mismanagement of project funds, which undermines the effectiveness of foreign aid (Dollar & Levin, 2005; Heyneman & Lee, 2016; Honig, 2018; Moyo, 2009). For instance, Miningou (2019) report that Sub-Saharan African (SSA) countries exhibit the lowest efficiency in converting educational aid into improved educational outcomes. This evidence shows the importance of sound management practices, reinforcing Heinzel and Liese (2021)'s argument that the World Bank must strategically reevaluate its organizational incentive structures. Specifically, this involves emphasizing management factors such as employing skilled and experienced TTLs, who have been shown to play a critical role in enhancing project performance (Denizer et al., 2013).

Furthermore, the findings reveal notable interactions between evaluated projects within the same vicinity, illustrating how the quality of educational aid impacts overall outcomes. HS projects within clusters are observed to reduce illiteracy rates when there are few or no other project types in the same area. In contrast, MS educational aid projects in urban areas require the existence of HS projects to achieve a reduction in illiteracy, whereas the combination of MS and HS projects in rural areas only increases illiteracy. The results highlight that, in urban areas, MS projects tend to perform better when part of a network of HS projects. This suggests that with less intensive management, projects might lack the resources or infrastructures necessary to achieve the desired impact. Strategic management appears to be key to maximizing the impact of educational aid and achieving sustainable improvements in educational development.

Additional evidence from my analysis reveals that the results are consistent across urban and rural

areas, aligning with claims about aid targeting (Briggs, 2018b; Öhler & Nunnenkamp, 2014; Öhler et al., 2019), which argue that the World Bank does not necessarily prioritize regional needs. The findings in my paper suggest that especially in rural areas, effective coordination between projects with different levels of management quality is critical. Policymakers should consider regional strategies that promote complementary project combinations and minimize detrimental overlaps. Moreover, when it comes to the magnitude of impact, there are no significant differences between urban and rural areas, but the effects are generally more pronounced in urban areas. A plausible explanation for this difference is that urban regions, despite being more developed, may still face significant educational challenges, particularly among marginalized groups. Educational aid targeting not only impoverished regions but also relatively affluent areas where vulnerable populations such as orphans remain excluded could address fundamental needs in both settings. Ultimately, the quality of project management outweighs the influence of geographic location on educational aid effectiveness. This reinforces the argument that management quality is crucial for accelerated progress in educational development, regardless of the contextual disparities between urban and rural areas.

7 Conclusion

In this paper, I have shown that, although other factors influence educational achievement in Africa, the management quality of World Bank-funded educational projects affects the illiteracy rate. This paper has contributed to the literature on World Bank project management effectiveness as it draws a connection between aid management quality and educational development. It has also expanded on the use of sub-national data to demonstrate that the quantity, monetary value, or location of educational aid is not the sole factor relevant for educational development in Africa but that management quality also plays an important role. Additionally, the findings of my research suggest the need for a more comprehensive methodology to measure the true impact of educational aid: that is, a measure that captures aid management quality, intensity, volume and duration.

Finally, the paper demonstrates that, despite significant improvements in educational development in Africa, management efficiencies can significantly accelerate educational development, underscoring the pivotal role of aid management in shaping educational outcomes in Africa. Good management practices among all stakeholders are essential. Enhancing the effectiveness of educational interventions can accelerate progress by improving enrollment rates, learning outcomes and educational infrastructures. To maximize the impact of educational assistance, the World Bank must prioritize the management quality of its operations when it comes to education. Through concerted efforts to improve aid management quality, donors can fulfill their mission of promoting inclusive and quality education for all. In conclusion, further research can enhance the measurement of educational outcome. Due to data limitations, my study adopted a proxy for illiteracy focusing on individuals with no formal education or those who had not completed primary education, under the assumption that these individuals may lack basic literacy skills if they never attended school or dropped out early. Although I was able to provide evidence of the effect of aid management quality on educational outcomes despite this constraint, future studies could benefit from incorporating more detailed educational metrics, such as disaggregated test scores or other comprehensive data, as they become available.

Asare

References

- AidData. (2017). WorldBank_GeocodedResearchRelease_Level1_v1.4.2 geocoded dataset. http://aiddata. org/research-datasets (accessed 31 March 2024).
- Asare, A. O., Dannemann, B. C., & Gören, E. (2023). Locust Infestations and Individual School Dropout: Evidence from Africa [University of Oldenburg, Department of Economics, Oldenburg]. Oldenburg Discussion Papers in Economics, (V-440-23).
- Ashton, L., Friedman, J., Goldemberg, D., Hussain, M. Z., Kenyon, T., Khan, A., & Zhou, M. (2022). A Puzzle with Missing Pieces: Explaining the Effectiveness of World Bank Development Projects. *The World Bank Research Observer*, 38(1), 115–146. https://doi.org/10.1093/wbro/lkac005
- Bedasso, B. (2024). Ministerial musical chairs: Does leadership turnover undermine the effectiveness of World Bank education aid? World Development Perspectives, 33, 100572. https://doi.org/10. 1016/j.wdp.2024.100572
- Bennell, P. (2002). Hitting the Target: Doubling Primary School Enrollments in Sub-Saharan Africa by 2015. World Development, 30(7), 1179–1194. https://doi.org/10.1016/s0305-750x(02)00027-x
- Birchler, K., & Michaelowa, K. (2016). Making aid work for education in developing countries: An analysis of aid effectiveness for primary education coverage and quality. *International Journal of Educational Development*, 48, 37–52. https://doi.org/10.1016/j.ijedudev.2015.11.008
- Bitzer, J., & Gören, E. (2024). The Impact of Foreign Aid on Local Development: A Grid Cell Analysis. The Journal of Development Studies, 60(10), 1557–1591. https://doi.org/10.1080/00220388.2024. 2377294
- Blattman, C., & Miguel, E. (2010). Civil War. Journal of Economic Literature, 48(1), 3–57. https://doi. org/10.1257/jel.48.1.3
- Briggs, R. C. (2018a). Leaving No One Behind? A New Test of Subnational Aid Targeting. Journal of International Development, 30(5), 904–910. https://doi.org/10.1002/jid.3357
- Briggs, R. C. (2018b). Poor targeting: A gridded spatial analysis of the degree to which aid reaches the poor in Africa. *World Development*, 103, 133–148. https://doi.org/10.1016/j.worlddev.2017.10.020
- Briggs, R. C. (2019). Results from single-donor analyses of project aid success seem to generalize pretty well across donors. *The Review of International Organizations*, 15(4), 947–963. https://doi.org/10.1007/s11558-019-09365-x
- Chauvet, L., Collier, P., & Duponchel, M. (2010). What Explains Aid Project Success in Post-Conflict Situations? https://doi.org/10.1037/e503962013-001
- Child, T. B. (2014). Hearts and minds cannot be bought: Ineffective reconstruction in Afghanistan. The Economics of Peace and Security Journal, 9(2). https://doi.org/10.15355/epsj.9.2.43

- CIESIN. (2018). Gridded Population of the World, Version 4.11 (GPWv4): Population Count, Revision 11 [Center for international earth science information network]. Columbia University. Palisades, NY: NASA Socioeconomic Data, Applications Center (SEDAC). https://sedac.ciesin.columbia. edu/data/set/gpw-v4-population-count-rev11 (accessed 01 April 2024).
- Clemens, M. A., Radelet, S., & Bhavnani, R. (2008). Counting Chickens When They Hatch: The Short Term Effect of Aid on Growth. SSRN Electronic Journal. https://doi.org/10.2139/ssrn.1112709
- Cockburn, J., & Dostie, B. (2007). Child Work and Schooling: The Role of Household Asset Profiles and Poverty in Rural Ethiopia. Journal of African Economies, 16(4), 519–563. https://doi.org/10. 1093/jae/ejl045
- Collier, P., & Hoeffler, A. (2006). Unintended Consequences: Does Aid Promote Arms Races?*. Oxford Bulletin of Economics and Statistics, 69(1), 1–27. https://doi.org/10.1111/j.1468-0084.2006. 00439.x
- d'Aiglepierre, R., & Wagner, L. (2013). Aid and Universal Primary Education. Economics of Education Review, 37, 95–112. https://doi.org/10.1016/j.econedurev.2013.09.001
- Davies, S., Pettersson, T., & Öberg, M. (2023). Organized violence 1989–2022, and the return of conflict between states. Journal of Peace Research, 60(4), 691–708. https://doi.org/10.1177/ 00223433231185169
- Denizer, C., Kaufmann, D., & Kraay, A. (2013). Good countries or good projects? Macro and micro correlates of World Bank project performance. *Journal of Development Economics*, 105, 288–302. https://doi.org/10.1016/j.jdeveco.2013.06.003
- DHS. (2024). GPS Data Collection. https://dhsprogram.com/Methodology/GPS-Data.cfm (accessed 07 January 2024).
- Doll, C. H., Muller, J.-P., & Elvidge, C. D. (2000). Night-time Imagery as a Tool for Global Mapping of Socioeconomic Parameters and Greenhouse Gas Emissions. AMBIO: A Journal of the Human Environment, 29(3), 157–162. https://doi.org/10.1579/0044-7447-29.3.157
- Dollar, D., & Levin, V. (2005, February). Sowing and Reaping: Institutional Quality and Project Outcomes in Developing Countries. The World Bank. https://doi.org/10.1596/1813-9450-3524
- Dollar, D., & Levin, V. (2006). The Increasing Selectivity of Foreign Aid, 1984–2003. *World Development*, 34(12), 2034–2046. https://doi.org/10.1016/j.worlddev.2006.06.002
- Dreher, A., Klasen, S., Vreeland, J. R., & Werker, E. (2013). The Costs of Favoritism: Is Politically Driven Aid Less Effective? *Economic Development and Cultural Change*, 62(1), 157–191. https: //doi.org/10.1086/671711

- Dreher, A., Nunnenkamp, P., & Thiele, R. (2008). Does Aid for Education Educate Children? Evidence from Panel Data. The World Bank Economic Review, 22(2), 291–314. https://doi.org/10.1093/ wber/lhn003
- Elvidge, C. D., Baugh, K. E., Kihn, E. A., Kroehl, H. W., Davis, E. R., & Davis, C. W. (1997). Relation between satellite observed visible-near infrared emissions, population, economic activity and electric power consumption. *International Journal of Remote Sensing*, 18(6), 1373–1379. https://doi.org/10.1080/014311697218485
- Findley, M., Young, J. K., Strandow, D., & Cat, O. (2023). Aiding War: Foreign Aid and the Intensity of Violent Armed Conflict. International Studies Quarterly, 67(3). https://doi.org/10.1093/isq/ sqad048
- Ghosh, T., Anderson, S., Powell, R., Sutton, P., & Elvidge, C. (2009). Estimation of Mexico's Informal Economy and Remittances Using Nighttime Imagery. *Remote Sensing*, 1(3), 418–444. https:// doi.org/10.3390/rs1030418
- Guillaumont, P., & Chauvet, L. (2001). Aid and Performance: A Reassessment. Journal of Development Studies, 37(6), 66–92. https://doi.org/10.1080/713601083
- Guillaumont, P., & Laajaj, R. (2006, October). When Instability Increases The Effectiveness Of Aid Projects. The World Bank. https://doi.org/10.1596/1813-9450-4034
- Heinzel, M., & Liese, A. (2021). Managing performance and winning trust: how World Bank staff shape recipient performance. The Review of International Organizations, 16(3), 625–653. https://doi. org/10.1007/s11558-021-09414-4
- Helskog, K. (2024). Effectiveness of aid on educational dropout in post-Genocide Rwanda. International Journal of Educational Development, 106, 103018. https://doi.org/10.1016/j.ijedudev.2024.103018
- Henderson, J. V., Storeygard, A., & Weil, D. N. (2012). Measuring Economic Growth from Outer Space. American Economic Review, 102(2), 994–1028. https://doi.org/10.1257/aer.102.2.994
- Heyneman, S. P., & Lee, B. (2016). International organizations and the future of education assistance. International Journal of Educational Development, 48, 9–22. https://doi.org/10.1016/j.ijedudev. 2015.11.009
- Honig, D. (2018). When Reporting Undermines Performance: The Costs of Politically Constrained Organizational Autonomy in Foreign Aid Implementation. International Organization, 73(1), 171–201. https://doi.org/10.1017/s002081831800036x
- Honig, D., Lall, R., & Parks, B. C. (2022). When Does Transparency Improve Institutional Performance? Evidence from 20,000 Projects in 183 Countries. American Journal of Political Science, 67(4), 1096–1116. https://doi.org/10.1111/ajps.12698

- Hsu, F.-C., Baugh, K., Ghosh, T., Zhizhin, M., & Elvidge, C. (2015). DMSP-OLS Radiance Calibrated Nighttime Lights Time Series with Intercalibration. *Remote Sensing*, 7(2), 1855–1876. https: //doi.org/10.3390/rs70201855
- ICF. (2024a). Demographic and Health Surveys (various) [Datasets]. Funded by USAID. Rockville, Maryland: ICF [Distributor]. https://www.dhsprogram.com/data/available-datasets.cfm (accessed 04 January 2024).
- ICF. (2024b). The DHS Program Spatial Data Repository [Funded by USAID]. https://www.dhsprogram. com/data/available-datasets.cfm (accessed 04 January2024).
- IEG. (2024). World Bank Project Ratings. https://ieg.worldbankgroup.org/ieg-data-world-bank-projectratings-and-lessons (accessed 1 January 2024).
- Ika, L. A., Diallo, A., & Thuillier, D. (2012). Critical success factors for World Bank projects: An empirical investigation. International Journal of Project Management, 30(1), 105–116. https://doi.org/10. 1016/j.ijproman.2011.03.005
- Kilby, C. (2000). Supervision and performance: the case of World Bank projects. Journal of Development Economics, 62(1), 233–259. https://doi.org/10.1016/s0304-3878(00)00082-1
- Lewin, K. M. (2020). Beyond business as usual: Aid and financing education in Sub Saharan Africa. International Journal of Educational Development, 78, 102247. https://doi.org/10.1016/j. ijedudev.2020.102247
- Limodio, N. (2021). Bureaucrat Allocation in the Public Sector: Evidence from the World Bank. *The Economic Journal*, 131 (639), 3012–3040. https://doi.org/10.1093/ej/ueab008
- Mahmud, M., & Riley, E. (2021). Household response to an extreme shock: Evidence on the immediate impact of the Covid-19 lockdown on economic outcomes and well-being in rural Uganda. World Development, 140, 105318. https://doi.org/10.1016/j.worlddev.2020.105318
- Marchetta, F., Sahn, D. E., & Tiberti, L. (2018). School or Work?: The Role of Weather Shocks in Madagascar. SSRN Electronic Journal. https://doi.org/10.2139/ssrn.3266899
- Maren, M. (2009). The Road to Hell. In *The ohio frontier* (pp. 95–119). Indiana University Press. https://doi.org/10.2307/j.ctt1zxz1gb.8
- Michaelowa, K., & Weber, A. (2008). Aid effectiveness in primary, secondary and tertiary education (research rep.). Background paper prepared for the Education for All Monitoring Report.
- Michaelowa, K. (2004). Aid Effectiveness Reconsidered: Panel Data Evidence for the Education Sector. SSRN Electronic Journal. https://doi.org/10.2139/ssrn.508382

- Miningou, E. W. (2019). Effectiveness of education aid revisited: Country-level inefficiencies matter. International Journal of Educational Development, 71, 102123. https://doi.org/10.1016/j.ijedudev. 2019.102123
- Moyo, D. (2009). Dead Aid: Why Aid is not Working and How There is a Better Way for Africa. London: Macmillan.
- Musah, A., Aawaar, G., & Nkansah, E. (2024). Role of institutional quality in the public education financing–educational quality nexus: evidence from Sub-Saharan Africa. Journal of Economics and Development, 26(3), 236–252. https://doi.org/10.1108/jed-07-2023-0133
- Nunnenkamp, P., Öhler, H., & Sosa Andrés, M. (2016). Need, Merit and Politics in Multilateral Aid Allocation: A District-level Analysis of World Bank Projects in India. *Review of Development Economics*, 21(1), 126–156. https://doi.org/10.1111/rode.12259
- Öhler, H., Negre, M., Smets, L., Massari, R., & Bogetić, Ž. (2019). Putting your money where your mouth is: Geographic targeting of World Bank projects to the bottom 40 percent (J. Paniagua, Ed.). *PLOS ONE*, 14(6), e0218671. https://doi.org/10.1371/journal.pone.0218671
- Öhler, H., & Nunnenkamp, P. (2014). Needs-Based Targeting or Favoritism? The Regional Allocation of Multilateral Aid within Recipient Countries. *Kyklos*, 67(3), 420–446. https://doi.org/10.1111/ kykl.12061
- Riddell, A., & Niño-Zarazúa, M. (2016). The effectiveness of foreign aid to education. International Journal of Educational Development, 48, 23–36. https://doi.org/10.1016/j.ijedudev.2015.11.013
- Song, T., Brazys, S., & Vadlamannati, K. C. (2020). Which Wheel Gets the Grease? Constituent Agency and Sub-national World Bank Aid Allocation. The Journal of Development Studies, 57(3), 519– 533. https://doi.org/10.1080/00220388.2020.1806247
- Stacy, B., Marin, S. V. M., Ciret, A., Rogers, H., & Cloutier, M.-H. (2023, May 25). What is driving the learning crisis? Clues from the initial rollout of the Global Education Policy Dashboard (W. B. Blogs, Ed.). Retrieved September 6, 2024, from https://blogs.worldbank.org/en/education/whatdriving-learning-crisis-clues-initial-rollout-global-education-policy-dashboard
- Stubbs, T. H., Kentikelenis, A. E., & King, L. P. (2016). Catalyzing Aid? The IMF and Donor Behavior in Aid Allocation. World Development, 78, 511–528. https://doi.org/10.1016/j.worlddev.2015.10.010
- Sundberg, R., & Melander, E. (2013). Introducing the UCDP Georeferenced Event Dataset. Journal of Peace Research, 50(4), 523–532. https://doi.org/10.1177/0022343313484347
- Thierry, M. A., & Emmanuel, O. N. B. (2022). Does Financial Development Increase Education Level? Empirical Evidence from Sub-Saharan Africa. Journal of the Knowledge Economy, 14(4), 3878– 3903. https://doi.org/10.1007/s13132-022-01020-y

- Trisko Darden, J. (2020, June). Aiding and Abetting: U.S. Foreign Assistance and State Violence. Stanford University Press. https://doi.org/10.1515/9781503611009
- UNESCO. (2024a). Education in Africa. https://uis.unesco.org/en/topic/education-africa (accessed 15 July 2024).
- UNESCO. (2024b). Education in Africa-Teachers. https://uis.unesco.org/en/topic/teachers (accessed 15 July 2024).
- UNICEF. (2024). Let me learn. Retrieved September 6, 2024, from https://www.unicef.org/learning-crisis
- Wood, R. M., & Molfino, E. (2016). Aiding Victims, Abetting Violence: The Influence of Humanitarian Aid on Violence Patterns During Civil Conflict. Journal of Global Security Studies, 1(3), 186–203. https://doi.org/10.1093/jogss/ogw007
- Wood, R. M., & Sullivan, C. (2015). Doing Harm by Doing Good? The Negative Externalities of Humanitarian Aid Provision during Civil Conflict. The Journal of Politics, 77(3), 736–748. https: //doi.org/10.1086/681239
- World Bank. (2024). Education in Eastern and Southern Africa. https://www.worldbank.org/en/region/ afr/brief/afe-education (accessed 22 July 2024).
- Yogo, T. U. (2017). Assessing the Effectiveness of Foreign Aid in the Education Sector in Africa: The Case of Primary Education: Foreign Aid in the Education Sector. African Development Review, 29(3), 389–402. https://doi.org/10.1111/1467-8268.12276
- Zamand, M., & Hyder, A. (2016). Impact of climatic shocks on child human capital: evidence from young lives data. *Environmental Hazards*, 15(3), 246–268. https://doi.org/10.1080/17477891.2016. 1185003
- Zürcher, C. (2019). The folly of "aid for stabilisation". *Third World Quarterly*, 40(5), 839–854. https://doi.org/10.1080/01436597.2019.1576519

A Tables

Statistics	Obs.	Mean	St. Dev.	Min	Max			
Disbursement	58,182	72,120,681	131,801,719	0	723,262,747			
Edu. disbursement	$58,\!182$	$6,\!838,\!527$	$26,\!549,\!611$	0	$415,\!563,\!490$			
Highly satisfactory edu. disb.	58,182	$2,\!575,\!795$	$10,\!637,\!138$	0	$128,\!650,\!425$			
Moderately satisfactory edu. disb	58,182	$3,\!552,\!316$	$22,\!678,\!475$	0	413,839,142			
Less satisfactory edu. disb	$58,\!182$	$380,\!821.1$	$4,\!433,\!806$	0	$85,\!965,\!303$			

Table A.1: Descriptive Statistics – Actual Disbursement

 $\it Notes:$ This table presents summary statistics of the actual disbursements of aid projects in U.S. dollars for the sample used in the analysis.

				Marginal Effects –			
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
ĤŚ	MS	ĹŚ	Freq.	Freq.(%)	Ν	Aarginal Effec	ts
					HS	MS	LS
			Pan	el A: Indicators			
0	1	0	3,170	43.95	_	0.98	_
1	1	0	1,353	18.76	-4.15	-3.18	—
1	0	0	1,205	16.71	0	_	—
0	1	1	486	6.74	_	-2.20	5.07
1	1	1	344	4.77	-6.90	-6.36	2.33
0	0	1	275	3.81	_	_	8.25
1	0	1	234	3.24	-2.74	_	5.50
0	2	0	89	1.23	_	0.98	_
0	2	2	20	0.28	_	-5.38	1.89
2	2	0	15	0.21	-8.31	-7.33	_
2	0	0	14	0.19	0	_	_
$\frac{2}{2}$	0	2	4	0.06	-5.49	_	2.76
0	0	2	2	0.03	_	_	8.25
0	3	0	1	0.01	_	0.98	_
0	3	3	1	0.01	_	-8.56	-1.29
			Pa	nel B: Counts			
0	1	0	1,425	19.76	_	0.38	
0	2	0	853	11.83	_	0.38	_
1	0	0	749	10.38	-0.71	_	_
0	3	0	331	4.59	_	0.38	_
1	1	0	323	4.48	-1.12	-0.04	_
0	6	0	267	3.70	_	0.38	_
2	0	0	264	3.66	-0.71	_	_
0	1	1	257	3.56	_	0.24	1.71
0	0	1	198	2.75	_	_	1.84
6	1	0	167	2.32	-1.12	-2.12	_
0	4	0	160	2.22	_	0.38	_
1	3	0	159	2.20	-1.96	-0.04	_
2	2	0	99	1.37	-1.54	-0.46	_
2	1	0	96	1.33	-1.12	-0.46	_
0	5	0	92	1.28	_	0.38	_
8	1	0	90	1.25	-1.12	-2.96	_
3	3	0	89	1.23	-1.96	-0.88	_
$\frac{1}{2}$	Õ	1	86	1.19	-0.71	_	1.84
3	Ő	$\overline{0}$	75	1.04	-0.71	_	_
				-			

Table A.2:	Marginal	Effects –	Urban
------------	----------	-----------	-------

Notes: This table presents the marginal effects for the urban sample. Panel A reports the marginal effects for indicators and Panel B for project counts.

Columns (1) to (3) present unique project quality combinations, while columns (4) to (5) show frequencies and percentages. Columns (6) to (8) display marginal effects calculated using Equations (5)–(7) and coefficients from Table 7, columns (3) and (6). For example, for a single MS project (row 1, Panel A, column (7)), substituting $HS_{dct} = 0$ and $LS_{dct} = 0$ into $0.9750 - 4.154 \times HS_{dct} - 3.178 \times LS_{dct}$ yields ≈ 0.98 percentage points.

Panel B calculates marginal effects using project counts and reports the marginal effects for project combinations appearing with $\geq 1\%$ frequency (166 unique combinations). For example, for a single HS project (Panel B, row 3, column (6)), substituting $MS_{dct} = 0$ and $LS_{dct} = 0$ into $-0.7080 - 0.4169 \times MS_{dct} - 0 \times LS_{dct}$ gives -0.71 percentage points as the marginal effect.

Insignificant coefficients are treated as zero. Bolded values are mentioned in the paper. Calculations use maximum decimals and are rounded to two decimal places.

			Table A.3:	Marginal Effects	– Rural		
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
ΗŚ	МŚ	(3)LS	Freq.	Freq.(%)	M	arginal Effec	ets
					HS	MS	LS
			Pan	el A: Indicators			
0	1	0	4,475	35.85	_	1.55	_
1	1	0	3,131	25.08	-3.6	-2.05	_
1	0	0	2,846	22.80	0	_	_
0	1	1	575	4.61	_	-3.40	4.26
1	1	1	542	4.34	-9.43	-7.00	-1.57
1	0	1	375	3.00	-5.83	_	3.39
0	0	1	345	2.76	_	_	9.21
0	2	0	102	0.82	_	1.55	_
0	2	2	34	0.27	_	-8.35	-0.69
	0	0	25	0.20	0	_	_
$2 \\ 2 \\ 2$	2	0	23	0.18	-7.20	-5.65	_
2	0	2	6	0.05	-11.66	_	-2.44
0	0	2	2	0.02	_	_	9.21
0	3	3	1	0.01	_	-13.31	-5.64
			Pa	anel B: Counts			
0	1	0	2,494	19.98	_	0.34	_
1	0	0	1,535	12.30	-0.33	_	_
0	2	0	1,006	8.06	_	0.34	_
1	1	0	816	6.54	-0.50	0.18	_
$2 \\ 2$	0	0	469	3.76	-0.33	—	_
2	1	0	384	3.08	-0.50	0.01	—
0	3	0	382	3.06	_	0.34	_
2	2	0	305	2.44	-0.66	0.01	—
0	0	1	289	2.32	_	_	4.33
0	1	1	273	2.19	_	0.03	4.02
3	0	0	255	2.04	-0.33	_	_
1	3	0	227	1.82	-0.83	0.18	_
0	4	0	225	1.80	_	0.34	—
4	0	0	207	1.66	-0.33	_	_
1	2	0	202	1.62	-0.66	0.18	_
0	5	0	199	1.59	_	0.34	—
0	6	0	199	1.59	_	0.34	—
2	0	1	141	1.13	-1.07	_	2.86

Table A.3	B: Margina	l Effects	– Rural
-----------	-------------------	-----------	---------

Notes: This table provides the marginal effects analysis for the rural sample, divided into two panels: Panel A presents results for indicator measure, while Panel B focuses on project counts.

Columns (1) to (3) describe unique project quality combinations, with columns (4) to (5) detailing their frequencies and percentages. Marginal effects are reported in columns (6) to (8), derived using Equations (5)–(7) and coefficients from Table 7, columns (3) and (6). For instance, the marginal effect of a single MS project in Panel A (row 1, column (7)) is computed as $1.548 - 3.600 \times HS_{dct} - 4.951 \times LS_{dct}$. Substituting $HS_{dct} = 0$ and $LS_{dct} = 0$ yields approximately 0.98 percentage points.

Panel B restricts the analysis to project counts with frequencies $\geq 1\%$ (208 unique combinations). For example, the marginal effect for a single HS project (Panel B, row, 2 column (6)) is calculated as $-0.3317 - 0.1660 \times MS_{dct} - 0.7359 \times LS_{dct}$, resulting in -0.33 percentage points when $MS_{dct} = 0$ and $LS_{dct} = 0$. Insignificant coefficients are assumed to be zero. Bolded marginal effects are mentioned in the paper. All calculations are performed with maximum decimals and rounded to two decimal places.

** • **	(1)	(2)	(3)	(4)	(5)	(6)
Variables	Educational aid	Management rating		Illiteracy (6-24 Educational aid		Interactions
		$nel A: 45 \ km \ buffer$			nel B: 40 km buffer	meeractions
Educational project	1.648***			1.158***		
HS	(0.2940)	-2.848***	-0.4824	(0.2886)	-3.132***	-1.183**
		(0.4390)	(0.5290)		(0.4230)	(0.5137)
MS		0.3075	1.809***		0.0701	1.352***
LS		$(0.3118) \\ 4.448^{***}$	$(0.3684) \\ 7.422^{***}$		$(0.3111) \\ 4.402^{***}$	$(0.3652) \\ 6.928^{***}$
		(0.5818)	(0.8469)		(0.6115)	(0.8647)
$HS \times MS$			-4.482^{***}			-3.868^{***}
$HS \times LS$			(0.5836) - 2.851^{***}			(0.5893) -2.015^{**}
			(0.9555)			(0.9999)
$MS \times LS$			-3.400^{***} (0.6853)			-3.279^{***} (0.7073)
	20.172	20.452	. ,	10.120	10 100	· /
Clusters with compl. edu. aid Clusters with compl. aid	$20,472 \\ 36,756$	$20,472 \\ 36,756$	$20,472 \\ 36,756$	$19,132 \\ 35,312$	$19,132 \\ 35,312$	$19,132 \\ 35,312$
Within \mathbb{R}^2	0.01067	0.01161	0.01344	0.00998	0.01130	0.01262
	Pa	nel C: 35km buffer		Pa	nel D: 30 km buffer	
Educational project	0.8353***			0.7349^{***}		
HS	(0.2808)	-3.097***	-1.434***	(0.2713)	-3.321***	-1.481***
п5		(0.4139)	(0.4993)		(0.4117)	(0.4871)
MS		0.0799	1.234^{***}		-0.1206	1.018***
		(0.3104)	(0.3592)		(0.3119)	(0.3578)
LS		3.457^{***} (0.6193)	5.736^{***} (0.8770)		4.130^{***} (0.6208)	6.216^{***} (0.8689)
$HS \times MS$		(0.0100)	-3.644***		(0.0200)	-4.255^{***}
			(0.6207)			(0.6535)
$HS \times LS$			-1.362 (1.011)			-1.730^{*} (1.009)
$MS \times LS$			-3.272^{***}			-2.725^{***}
			(0.7367)			(0.7545)
Clusters with compl. edu. aid	17,672	17,672	17,672	15,967	15,967	15,967
Clusters with compl. aid	33,648	33,648	0.01153	0.00863	0.01003	0.01113
		nel E: 25 km buffer			nel F: 20 km buffer	
Educational project	0.6874^{**} (0.2677)			0.7589^{***} (0.2679)		
HS	(0.2011)	-3.209***	-1.408***	(0.2019)	-2.991***	-0.9729**
		(0.4165)	(0.4837)		(0.4196)	(0.4867)
MS		$0.1696 \\ (0.3197)$	1.274^{***}		0.3194	1.328^{***}
LS		(0.3197) 3.973***	$(0.3644) \\ 5.396^{***}$		(0.3311) 3.799^{***}	$(0.3754) \\ 5.390^{***}$
		(0.6543)	(0.8987)		(0.6979)	(0.9338)
$\text{HS} \times \text{MS}$			-4.843^{***} (0.6964)			-5.424^{***} (0.7485)
$HS \times LS$			(0.0904) -0.7078			(0.7485) -2.360^{**}
			(1.033)			(1.070)
$MS \times LS$			-2.254^{***}			-1.769^{**}
<u> </u>	10.000	10.000	(0.7966)	11 000	11 000	(0.8183)
Clusters with compl. edu. aid Clusters with compl. aid	$13,908 \\ 29,291$	$13,908 \\ 29,291$	$13,908 \\ 29,291$	$11,620 \\ 26,055$	$11,620 \\ 26.055$	$11,620 \\ 26,055$
Within \mathbb{R}^2	0.00795	0.00914	0.01013	0.00739	0.00828	0.00924
Year FE	Yes	Yes	Yes	Yes	Yes	Yes
District FE	Yes	Yes	Yes	Yes	Yes	Yes
Baseline controls	Yes	Yes	Yes	Yes	Yes	Yes
Observations	58,182	58,182	58,182	58,182	58,182	58,182

 Table A.4: Replication of Baseline Results at Different Buffer Sizes – Indicators

Notes: The dependent variable is the share of individuals between the ages of 6 and 24 years who are illiterate (see Section 2 for definition of illiteracy) multiplied by 100. Each column comes from a unique regression. Base line controls include: geographic controls (number of conflicts 5 years prior to the survey date, population per 100,000, and mean nighttime light); weather (mean precipitation, mean temperature, and mean drought) and DHS cluster controls (average age in the cluster, average household size, and the share of sons and daughters) in the cluster. All regressions include year and district (that is, administrative level 2) FE. Clustered standard errors by DHS clusters in parentheses. *Signif. Codes:* ***: 0.01, **: 0.05, *: 0.1

	(1)	(2)	(3)	(4)	(5)	(6)
Variables	Educational aid			Illiteracy (6-24 Educational aid	l years) l Management rating	Interactions
		$nel A: 45 \ km \ buffer$			anel B: 40 km buffer	moractions
Num. educational projects	0.0901***			0.0794^{**}		
HS	(0.0299)	-0.8391***	-0.4503***	(0.0330)	-1.025***	-0.6799***
MG		(0.1383)	(0.1536)		(0.1499)	(0.1672)
MS		0.3298^{***} (0.0887)	0.5536^{***} (0.1024)		0.3268^{***} (0.0973)	0.5115^{***} (0.1121)
LS		2.111***	2.592***		2.107^{***}	2.469***
$\mathrm{HS} \times \mathrm{MS}$		(0.2681)	(0.3309) - 0.3640^{***}		(0.2854)	(0.3494) - 0.3737^{***}
			(0.0615)			(0.0789)
$HS \times LS$			-0.2641 (0.2210)			-0.2307 (0.2484)
$\rm MS\timesLS$			-0.1279^{***}			-0.0925^{*}
			(0.0481)			(0.0545)
Clusters with compl. edu. aid	20,472	20,472	20,472	19,132	19,132	19,132
Clusters with compl. aid Within \mathbb{R}^2	$36,756 \\ 0.01010$	$36,756 \\ 0.01156$	$36,756 \\ 0.01210$	$35,312 \\ 0.00971$	$35,312 \\ 0.01117$	$35,312 \\ 0.01151$
		nel C: 35km buffer	-		anel D: 30 km buffer	-
Num. educational projects	0.0741**			0.0955^{**}		
HS	(0.0355)	-1.138***	-0.7402***	(0.0392)	-1.225***	-0.6878***
		(0.1634)	(0.1847)		(0.1858)	(0.2084)
MS		0.3622^{***} (0.1052)	0.5426^{***} (0.1201)		0.3342^{***} (0.1165)	0.5184^{***} (0.1323)
LS		1.887^{***}	2.187^{***}		2.015***	(0.1020) 2.277^{***}
$\mathrm{HS} \times \mathrm{MS}$		(0.2993)	(0.3670) - 0.5163^{***}		(0.3044)	(0.3763) - 0.7922^{***}
115 × 1415			(0.1013)			(0.1253)
$\mathrm{HS} \times \mathrm{LS}$			-0.1823			-0.2090
$MS \times LS$			$(0.2750) \\ -0.0645$			$(0.3094) \\ -0.0249$
			(0.0612)			(0.0706)
Clusters with compl. edu. aid	17,672	17,672	17,672	15,967	15,967	15,967
Clusters with compl. aid Within \mathbb{R}^2	$33,648 \\ 0.00918$	$33,648 \\ 0.01044$	$33,648 \\ 0.01078$	$31,701 \\ 0.00855$	$31,701 \\ 0.00970$	$31,701 \\ 0.01012$
		nel E: 25 km buffer	0.01010		Panel F: 20 km buffer	0.01012
Num. educational projects	0.1261***	nei D. 20 km bajjer		0.1562^{***}	anei 1. 20 km bajjer	
HS	(0.0449)	-1.329***	-0.7929***	(0.0526)	-1.475***	-0.6727***
115		(0.2038)	(0.2348)		(0.2182)	(0.2513)
MS		0.4843^{***}	0.6687^{***}		0.5984^{***}	0.8380^{***}
LS		(0.1388) 2.059^{***}	(0.1556) 2.169^{***}		$(0.1655) \\ 2.156^{***}$	(0.1852) 2.353^{***}
		(0.3333)	(0.3715)		(0.3730)	(0.4076)
$\mathrm{HS} \times \mathrm{MS}$			-0.9777^{***} (0.1739)			-1.562^{***} (0.2196)
HS \times LS			-0.0263			-0.5635
$MS \times LS$			(0.3285) -0.0082			$(0.3913) \\ 0.0068$
			(0.0801)			(0.0990)
Clusters with compl. edu. aid	13,908	13,908	13,908	11,620	11,620	11,620
Clusters with compl. aid Within \mathbb{R}^2	29,291	29,291	$29,291 \\ 0.00934$	26,055	26,055	26,055
	0.00791	0.00903		0.00732	0.00839	0.00884
Year FE District FE	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes
Baseline controls Observations	Yes 58,182	Yes	Yes	Yes	Yes	Yes 58,182
	58 189	58,182	58,182	58,182	58,182	

Table A.5:	Replication	of Baseline	Results at	Different	Buffer	Sizes –	Project C	ounts
------------	-------------	-------------	------------	-----------	--------	---------	-----------	-------

Notes: The dependent variable is the share of individuals between the ages of 6 and 24 years who are illiterate (see Section 2 for definition of illiteracy) multiplied by 100. Each column comes from a unique regression. Baseline controls include: geographic controls (number of conflicts 5 years prior to the survey date, population per 100,000, and mean nightime light); weather (mean precipitation, mean temperature, and mean drought) and DHS cluster controls (average age in the cluster, average household size, and the share of sons and daughters) in the cluster. All regressions include year and district (that is, administrative level 2) FE. Clustered standard errors by DHS clusters in parentheses. *Signif. Codes:* ***: 0.01, **: 0.05, *: 0.1

Variables	(1)	(2) Depender	(3) nt Variable:	(4) Illiteracy (6-	(5) 24 years)	(6)
variables		Management rating	Interactions		aid Management rating	Interactions
ln(Value edu.)	$Pa = 0.0761^{***}$	nel A: 45 km buffer		0.0498***	Panel B: 40 km buffer	
in(value edu.)	(0.0152)			(0.0150)		
$\ln(HS)$		-0.1427^{***}	-0.1111^{***}	. ,	-0.1569***	-0.1294^{***}
$\ln(MS)$		$(0.0216) \\ 0.0194 \\ (0.0156)$	$(0.0230) \\ 0.0090 \\ (0.0160)$		$(0.0209) \\ 0.0079 \\ (0.0156)$	(0.0224) -0.0059 (0.0160)
$\ln(LS)$		0.2273^{***}	0.3097^{***}		0.2275^{***}	0.2955^{***}
$\ln(\text{HS} \times \text{MS})$		(0.0311)	(0.0356) - 0.0105^{***} (0.0014)		(0.0327)	(0.0368) - 0.0091^{***} (0.0014)
$\ln(\text{HS} \times \text{LS})$			-0.0074***			-0.0053**
$\ln(MS \times LS)$			$\begin{array}{c}(0.0024)\\-0.0079^{***}\\(0.0018)\end{array}$			$(0.0025) \\ -0.0076^{***} \\ (0.0018)$
Clusters with compl. edu. aid	20,472	20,472	20,472	19,132	19,132	19,132
Clusters with compl. aid Within \mathbb{R}^2	$36,756 \\ 0.01053$	$36,756 \\ 0.01157$	$36,756 \\ 0.01338$	$35,312 \\ 0.00988$	$35,312 \\ 0.01127$	$35,312 \\ 0.01255$
WITHIN IC		inel C: 35km buffer	0.01000	0.00500	Panel D: 30 km buffer	0.01200
ln(Value edu.)	0.0302**	iner C. Sokin bujjer		0.0221	1 anei D. 50 km bajjer	
$\ln(\text{HS})$	(0.0146)	-0.1555***	-0.1310***	(0.0142)	-0.1664***	-0.1447***
		(0.0206)	(0.0218)		(0.0205)	(0.0214)
$\ln(MS)$		$\begin{array}{c} 0.0095\\ (0.0156) \end{array}$	-0.0085 (0.0158)		-0.0012 (0.0157)	-0.0203 (0.0157)
$\ln(LS)$		0.1771^{***}	0.2376^{***}		0.2094^{***}	0.2644^{***}
$\ln(\text{HS} \times \text{MS})$		(0.0332)	(0.0372) - 0.0087^{***}		(0.0331)	(0.0368) - 0.0102^{***}
			(0.0015)			(0.0016)
$\ln(\text{HS} \times \text{LS})$			-0.0034 (0.0025)			-0.0043^{*} (0.0025)
$\ln(MS \times LS)$			-0.0076* ^{**}			-0.0061* ^{**}
			(0.0019)			(0.0020)
Clusters with compl. edu. aid Clusters with compl. aid	$17,672 \\ 33,648$	$17,672 \\ 33,648$	$17,672 \\ 33,648$	$15,967 \\ 31,701$	$15,967 \\ 31,701$	$15,967 \\ 31,701$
Within \mathbb{R}^2	0.00922	0.01043	0.01148	0.00854	0.00999	0.01105
		nel E: 25 km buffer			Panel F: 20 km buffer	
ln(Value edu.)	0.0238^{*} (0.0141)			0.0299^{**} (0.0141)		
$\ln(HS)$	(0.0141)	-0.1610***	-0.1377***	(0.0141)	-0.1506***	-0.1426***
$\ln(MS)$		$(0.0207) \\ 0.0145$	(0.0214) -0.0072		$(0.0209) \\ 0.0228$	(0.0213) -0.0040
III(1013)		(0.0145) (0.0161)	(0.0161)		(0.0167)	(0.0165)
$\ln(LS)$		0.2020^{***}	0.2386^{***}		0.1929^{***}	0.2291^{***} (0.0399)
$\ln(\text{HS} \times \text{MS})$		(0.0347)	(0.0380) - 0.0117^{***}		(0.0370)	-0.0130***
$\ln(\text{HS} \times \text{LS})$			(0.0017) -0.0022			(0.0018) - 0.0068^{**}
`			(0.0026)			(0.0027)
$\ln(MS \times LS)$			-0.0047^{**} (0.0021)			-0.0033 (0.0021)
Clusters with compl. edu. aid	13,908	13,908	13,908	11,620	11,620	11,620
Clusters with compl. aid Within \mathbb{R}^2	$29,291 \\ 0.00788$	$29,291 \\ 0.00912$	$29,291 \\ 0.01007$	$26,055 \\ 0.00732$	$26,055 \\ 0.00827$	$26,055 \\ 0.00921$
Year FE	Yes	Yes	Yes	Yes	Yes	Yes
	3.7	Yes	Yes	Yes	Yes	Yes
District FE Baseline controls	Yes Yes	Yes	Yes	Yes	Yes	Yes

Notes: The dependent variable is the share of individuals between the ages of 6 and 24 years who are illiterate (see Section 2 for definition of illiteracy) multiplied by 100. All disbursement are logged transformed using $\ln(0.01+..)$. Each column comes from a unique regression. Baseline controls include: geographic controls (number of conflicts 5 years prior to the survey date, population per 100,000, and mean nighttime light); weather (mean precipitation, mean temperature, and mean drought) and DHS cluster controls (average age in the cluster, average household size, and the share of sons and daughters) in the cluster. All regressions include year and district (that is, administrative level 2) FE. Clustered standard errors by DHS clusters in parentheses. *Signif. Codes:* ***: 0.01, **: 0.05, *: 0.1

Variable	Description
Illiteracy	The variable is constructed using DHS variables "schooling status (hv129)"; "whether the household member is
	still in school" (hv110); whether the "household member attended school during current school year (hv121)";
	"educational attainment (hv109)" and "highest educational level attained (hv106)". It is then aggregated to
	the cluster level and then the shares are calculated from it.
Average Age	Age of each household member ("based on the reported age of the individual (hv105")) and then take the
	average age in the cluster.
Average Household size	Equal to the number of persons in each household and then taking the average within the cluster (based on
	"number of household members (hv009)").
Son or Daughter	An indicator equals one if the member of the household is a son or daughter of the head. It is based on the
	"relationship to head (hv101)". Equals one if hv101 is equal to three.

Table A.7: Description of DHS Variables

Notes: The table gives a detailed description of all the DHS variables used in the analysis. The variables can be found in the household record (PR) of the DHS program data types. Refer to supplementary data for how variables are computed.

	Table A.8: DHS Survey Version/Year DHS Version/Year					
Countries	Year	Version (\dot{PR})	Version (GE)			
Angola	2006	51	52			
Angola	2011	62	61			
Angola	2015	71	71			
Burkina Faso	1993	21	23			
Burkina Faso	1999	31	32			
Burkina Faso	2003	44	43			
Burkina Faso	2010	62	61			
Burkina Faso	2014	71	71			
Benin	1996	31	33			
Benin	2001	41	42			
Benin	2012	61	61			
Benin	2017	71	71			
Burundi	2010	61	61			
Burundi	2012	6Å	6Å			
Burundi	2016	71	71			
Democratic Republic of the Congo	2007	51	52			
Democratic Republic of the Congo	2013	61	61			
Central African Republic	1994	31	33			
Côte d'Ivoire	1994	35	33			
Côte d'Ivoire	1998	3Ă	3B			
Côte d'Ivoire	2012	62	61			
Cameroon	2012	45	42			
Cameroon	2001	61	61			
Cameroon	2011	71	71			
Egypt	1992	21	22			
Egypt	1995	33	32			
Egypt	2000	42	42			
Egypt	2003	4Å	4B			
Egypt	2005	51	52			
Egypt	2008	5Å	<u>5D</u>			
Egypt	2014	61	61			
Ethiopia	2000	41	42			
Ethiopia	2005	51	52			
Ethiopia	2010	61	61			
Ethiopia	2010	71	71			
Ethiopia	2010	81	81			
Gabon	2013	61	61			
Gabon	2012	71	71			
Ghana	1993	31	33			
Ghana	$\frac{1993}{1998}$	41	$\frac{53}{42}$			
Ghana	2003	41 4B	$\frac{42}{4B}$			
Ghana	2003	4D 5A	4D 5A			
Ghana	2008	72	71			
	2014 2016	72 7B	7A			
Ghana Chana	2016	<u>7B</u> 82	<u>/A</u> 81			
Ghana						
Gambia Notes: The table shows country, version, a	2019	81	81			

Notes: The table shows country, version, and year of the DHS survey included in the sample. PR is household recode dataset and GE is the GPS dataset for cluster locations.

a		DHS Version/Year	
Countries	Year	Version (PR)	Version (GE)
Guinea	1999	41	42
Guinea	2005	53	52
Guinea	2012	62	61
Guinea	2018	71	71
Kenya	2003	42	43
Tanzania	2008	52	52
Kenya	2014	72	71
Kenya	2015	7A	7A
Kenya	2020	81	81
Liberia	2007	51	52
Liberia	2009	5A	$5\mathrm{C}$
Liberia	2011	61	61
Liberia	2013	6Å	6A
Liberia	2016	71	71
Liberia	2010	7Å	7Å
Lesotho	2013	41	42
Lesotho	2004	61	62
Lesotho	2009	71	71
Western Sahara	2014 2003	43	43
			$\frac{43}{32}$
Madagascar	1997	31	
Madagascar	2008	51	53
Madagascar	2011	61	61
Madagascar	2013	6A	6A
Madagascar	2016	71	71
Malawi	2000	41	43
Malawi	2004	$4\mathrm{E}$	$4\mathrm{B}$
Malawi	2010	61	62
Malawi	2012	6A	6A
Malawi	2014	72	71
Malawi	2015	7A	7A
Malawi	2017	7I	7I
Mozambique	2009	51	52
Mozambique	2011	62	61
Mozambique	2015	71	71
Mozambique	2018	7Å	7Å
Vigeria	2003	4C	4B
Nigeria	2003	53	52
Vigeria	2000	61	61
Vigeria	2010	6A	6A
Vigeria	2013	71	71
Nigeria	2013	71 7B	71 7B
Niger	1992	22	23
Niger	<u>1992</u> 1998	31	$\frac{23}{32}$
	2012		
Niger		61	61
Rwanda	2005	53	54 DD is househ

A.8 Continued

wanda20055354Notes: The table shows country, version, and year of the DHS survey included in the sample. PR is householdrecode dataset and GE is the GPS dataset for cluster locations.

	A.8 Continued	DHS Version/Year	
Countries	Year	Version (PR)	Version (GE)
Rwanda	2008	$\frac{1}{5A}$	5B
Rwanda	2008	61	61
Rwanda	2010	70	$\frac{01}{72}$
Rwanda	2014 2019	81	81
Sierra Leone	2019	51	53
	2008	61	<u> </u>
Sierra Leone	2013	73	
Sierra Leone	2010		71
Sierra Leone		7A	7A
Senegal	1993	21	23
Senegal	1997	32	32
Senegal	2005	4A	4B
Senegal	2008	5A	5A
Senegal	2010	61	61
Senegal	2012	6D	6A
Senegal	2015	$7\mathrm{H}$	7A
Senegal	2017	7Z	$7\mathrm{R}$
Senegal	2018	81	81
Senegal	2019	8B	8B
Senegal	2020	8I	8I
Swaziland	2006	52	53
Chad	2014	71	71
Togo	1998	31	32
Togo	2013	61	62
Togo	2017	71	71
Tanzania	1999	41	43
Tanzania	2007	51	52
Tanzania	2010	63	61
Tanzania	2012	60 6A	6A
Tanzania	2012	7B	7A
Tanzania	2017	7I	71
Uganda	2000	41	43
Uganda	2000	52	53
Uganda	2000	5A	55 5A
Uganda	2009	<u> </u>	<u> </u>
Uganda	2011 2010	6A 61	6A 61
Uganda	2010	0A 72	71
	2014 2016		
Uganda		7B 7I	7A 7I
Uganda	2018	7I 71	7I 71
South Africa	2017	71	71
Zambia	2007	51	52
Zambia	2013	61	61
Zambia	2018	71	71
Zimbabwe	1999	42	42
Zimbabwe	2005	52	52
Zimbabwe	2010	62	61
Zimbabwe	2015	72	72

A.8 Continued

Zimbabwe20157272Notes: The table shows country, version, and year of the DHS survey included in the sample. PR is householdrecode dataset and GE is the GPS dataset for cluster locations.

Zuletzt erschienen /previous publications:

- V-449-25 Abigail O. Asare, More Than Money: The Critical Role of Management in Educational Aid Effectiveness in Africa
- V-448-24 Jürgen Bitzer, Bernhard C. Dannemann, Erkan Gören, Heterogeneous Effects of Foreign Aid on Local Economic Development
- V-447-24 **Heinz Welsch**, Explaining Happiness Trends in Europe by Welfare Policies and Economic Growth: Easterlin and O'Connor Revisited
- V-446-24 **Abigail Asare, Laura Schürer**, Incidence of Carbon Pricing in Tanzania: Using Revenues to Empower Low-Income Households with Renewable Energy
- V-445-24 **Heinz Welsch**, Are National Climate Change Mitigation Pledges Shaped by Citizens' Climate Action Preferences? Evidence from Globally Representative Data
- V-444-24 **Heinz Welsch**, Household Sector Carbon Pricing, Revenue Rebating, and Subjective Well-Being: A Dollar is not a Dollar
- V-443-24 **Heinz Welsch**, Do National Well-Being Scores Capture Nations' Ecological Resilience? Evidence for 124 Countries
- V-442-23 **Jürgen Bitzer, Erkan Gören, Heinz Welsch,** How the Well-Being Function Varies with Age: The Importance of Income, Health, and Social Relations over the Life Cycle
- V-441-23 **Heinz Welsch,** Why is Satisfaction from Pro-Environmental Behaviors Increasing in Costs? Insights from the Rational-Choice Decision-Error Framework
- V-440-23 Abigail O. Asare, Bernhard C. Dannemann, Erkan Gören, Locust Infestations and Individual School Dropout: Evidence from Africa
- V-439-22 Christoph Böhringer, Carolyn Fischer, Nicholas Rivers, Intensity-Based Rebating of Emission Pricing Revenues
- V-438-21 **Heinz Welsch,** What Shapes Cognitions of Climate Change in Europe? Ideology, Morality and the Role of Educational Attainment
- V-437-21 **Heinz Welsch**, Do Social Norms Trump Rational Choice in Voluntary Climate Change Mitigation? Multi-Country Evidence of Social Tipping Points
- V-436-21 **Emmanuel Asane-Otoo, Bernhard C. Dannemann**, Station heterogeneity and asymmetric gasoline price responses
- V-435-21 Christoph Böhringer, Thomas F. Rutherford, Jan Schneider, The Incidence of CO₂ Emission Pricing Under Alternative International Market Responses
- V-434-21 Christoph Böhringer, Sonja Peterson, Thomas F. Rutherford, Jan Schneider, Malte Winkler, Climate Policies after Paris: Pledge, Trade and Recycle
- V-433-20 **Bernhard C. Dannemann**, Better Off On Their Own? How Peer Effects Determine International Patterns of the Mathematics Gender Achievement Gap
- V-432-20 Christoph Böhringer, Carolyn Fischer, Kill Bill or Tax: An Analysis of Alternative CO2 Price Floor Options for EU Member States
- V-431-20 **Heinz Welsch**, How Climate-Friendly Behavior Relates to Moral Identity and Identity-Protective Cognition: Evidence from the European Social Surveys
- V-430-20 Christoph Böhringer, Knut Einar Rosendahl, Europe beyond Coal An Economic and Climate Impact Assessment
- V-429-20 **Oliver Richters**, Modeling the out-of-equilibrium dynamics of bounded rationality and economic constraints
- V-428-20 **Bernhard C. Dannemann,** Peer Effects in Secondary Education: Evidence from the 2015 Trends in Mathematics and Science Study Based on Homophily
- V-427-19 Christoph Böhringer, Knut Einar Rosendahl, Halvor Briseid Storrøsten, Smart hedging against carbon leakage

- V-426-19 **Emmanuel Asane-Otoo, Bernhard Dannemann,** Rockets and Feathers Revisited: Asymmetric Retail Fuel Pricing in the Era of Market Transparency
- V-425-19 **Heinz Welsch,** Moral Foundations and Voluntary Public Good Provison: The Case of Climate Change
- V-424-19 Gökçe Akın-Olçum, Christoph Böhringer, Thomas Rutherford, Andrew Schreiber, Economic and Environmental Impacts of a Carbon Adder in New York
- V-423-19 Jasper N. Meya, Paul Neetzow, Renewable energy policies in federal government systems
- V-422-19 **Philipp Biermann, Heinz Welsch**, Changing Conditions, Persistent Mentality: An Anatomy of East German Unhappiness, 1990-2016
- V-421-19 **Philipp Biermann, Jürgen Bitzer, Erkan Gören,** The Relationship between Age and Subjective Well-Being: Estimating Within and Between Effects Simultaneously
- V-420-19 **Philipp Poppitz,** Multidimensional Inequality and Divergence: The Eurozone Crisis in Retrospect
- V-419-19 **Heinz Welsch**, Utilitarian and Ideological Determinants of Attitudes toward Immigration: Germany before and after the "Refugee Crisis"
- V-418-19 Christoph Böhringer, Xaquin Garcia-Muros, Mikel González-Eguino, Greener and Fairer: A Progressive Environmental Tax Reform for Spain
- V-417-19 Heinz Welsch, Martin Binder, Ann-Kathrin Blankenberg, Pro-environmental norms and subjective well-being: panel evidence from the UK